LAW OFFICE OF DANIEL J. BERMAN Suite 801, DNA Building 238 Archbishop Flores Street Hagåtña, Guam 96910 Telephone No.: (671) 922-7570

Email: djberman@pacificlawyers.law

Attorneys for Plaintiff: RELYANT GLOBAL, LLC

UNITED STATES DISTRICT COURT

FOR THE DISTRICT OF GUAM

UNITED STATES of AMERICA for the use and benefit of RELYANT GLOBAL, LLC,	Civil Action No. 1:25-cv-00032
Plaintiff,)
vs.)
TRAVELERS CASUALTY AND SURETY COMPANY OF AMERICA, THE CONTINENTAL INSURANCE COMPANY, FEDERAL INSURANCE COMPANY, ZURICH AMERICAN INSURANCE COMPANY, and SAFECO INSURANCE COMPANY OF AMERICA,	
Defendants.)))

COMPLAINT

COMES NOW Plaintiff United States of America for the use and benefit of Relyant Global, LLC, by and through its undersigned attorneys, and for its cause of action against the Defendants, Travelers Casualty and Surety Company of America, The Continental Insurance Company, Federal Insurance Company, Zurich American Insurance Company, and Safeco Insurance Company of America ("Defendants"), states as follows:

PARTIES

- Plaintiff Relyant Global, LLC ("Relyant") is, and at all times relevant has been,
 a Tennessee limited liability company with its principal office and place of business located at
 335 High Street, Maryville, Blount County, Tennessee.
- 2. Defendant Travelers Casualty and Surety Company of America ("Travelers") is a foreign insurer organized under the laws of the State of Connecticut and authorized to issue contract surety bonds in the Territory of Guam. Travelers may be served with process at its statutory home office address: One Tower Square, Hartford, CT 06183.
- 3. Defendant The Continental Insurance Company ("Continental") is a foreign insurer organized under the laws of the State of Pennsylvania and authorized to issue contract surety bonds in the Territory of Guam. Continental may be served with process at its statutory home office address: 151 N. Franklin Street, Chicago, IL 60606.
- 4. Defendant Federal Insurance Company ("Federal") is a foreign insurer organized under the laws of the State of Indiana and authorized to issue contract surety bonds in the Territory of Guam. Federal may be served with process at its statutory home office address: 202B Hall's Mill Road, Whitehouse Station, NJ 08889.
- 5. Defendant Zurich American Insurance Company ("Zurich") is a foreign insurer organized under the laws of the State of New York and authorized to issue contract surety bonds in the Territory of Guam. Zurich may be served with process at its statutory home office address: 1400 American Lane, Schaumburg, IL 60196.
- 6. Defendant Safeco Insurance Company of America ("Safeco") is a foreign insurer organized under the laws of the State of New Hampshire and authorized to issue contract surety bonds in the Territory of Guam. Safeco may be served with process at its statutory home office address: 175 Berkeley Street, Boston, MA 02116.

JURISDICTION AND VENUE

- 7. This Court has original subject matter jurisdiction over this action pursuant to 28 U.S.C. § 1352 in that this matter involves an action on a payment bond executed pursuant to 40 U.S.C. § 3131 et. seq. (the "Miller Act").
- 8. Venue is proper in this judicial district because Plaintiff's cause of action arose in this district and the construction project at issue is located in this district.

FACTUAL BACKGROUND

- 9. In or about 2017, Granite Obayashi, a Joint Venture ("GOJV"), entered into Contract Number N62742-17-C-1324 (the "Prime Contract") with Naval Facilities and Engineering Command Pacific ("NAVFAC" or the "Government") to perform work as the prime contractor on the J-001B Finegayan Utilities and Site Improvements Phase I project (the "Project").
- 10. As required by the provisions of the Prime Contract and applicable federal law, GOJV provided a payment bond for the Project from Defendants (the "Payment Bond"). The Payment Bond was issued with GOJV, as principal, and with Defendants, jointly and severally liable, as sureties, for the benefit of subcontractors and material suppliers, such as Relyant. A copy of the Payment Bond is attached hereto as Exhibit A.
- 11. On October 1, 2020, Relyant entered into Subcontract No. 08.0040 with GOJV to provide munitions and explosives of concern ("MEC") clearance services on the Project (the "Subcontract"). A copy of the Subcontract is attached hereto as Exhibit B.
- 12. International Consolidated Contracting, LLC ("ICC") was originally GOJV's MEC subcontractor for the Project. However, after ICC ceased work on the Project, GOJV later asked Relyant to submit (and Relyant did submit) a proposal to GOJV to perform MEC work on the

Project in Areas A, B, D, and F, in reference to Prime Contract change order "PCO-42." GOJV awarded this work to Relyant, and GOJV and Relyant entered into the Subcontract.

- 13. Under the Subcontract, Relyant's scope of work was limited to performing MEC clearance on areas A, B, D, and F of the Project. However, through change orders, Relyant also performed some additional work in other limited areas of the Project site at the direction of GOJV.
- 14. Relyant conducted fieldwork on the Project between November 13, 2020, and November 4, 2022, and provided additional MEC work and construction support between March 20, 2023, and May 4, 2023. Most of Relyant's scope of work on the Project is now complete, but additional work, including completion of after-action reports comprehensively detailing Relyant's MEC work on the Project, has not been completed. Relyant's after-action reports are part of Relyant's scope of work under its Subcontract with GOJV and part of GOJV's scope of work under the Prime Contract.
- 15. Relyant previously submitted its after-action reports to GOJV, and, upon information and belief, GOJV has submitted Relyant's after-action reports to the Government. However, the after-action reports have not yet been approved by the Government. Because the Government will require Relyant to make revisions to the after-action reports before they are accepted, Relyant's scope of work on the Project is not complete. Accordingly, Relyant's cause of action under the Miller Act has not yet accrued for statute of limitations purposes.
- 16. Relyant's Invoice # 20147-26 in the amount of \$100,181.51 for work Relyant performed pursuant to the Subcontract remains unpaid. A copy of Relyant's Invoice # 20147-26 is attached hereto as Exhibit C.
- 17. In addition, on February 24, 2023, Relyant submitted two Requests for Equitable Adjustment ("REAs") to GOJV pursuant to Section 14.4 of the Subcontract, for submission by

GOJV to the Government. The first REA, in the amount of \$1,658,036.77 (later revised to \$1,481,425.09 in an amended submission), is based on soil piles encountered which differed materially from the contract documents in that they averaged over 7,779 anomalies per acre despite the stated maximum anomaly count of 1,800 per acre in the contract documents and in the Government's Explosive Safety Submission ("ESS"), version 6. A copy of Relyant's first REA is attached hereto as Exhibit D.

- 18. Similarly, Relyant's second REA, in the amount of \$1,101,355.53, is based on trash pits Relyant encountered which differed materially from the contract documents in that they averaged over 14,791 anomalies per acre despite the stated maximum anomaly count of 1,800 per acre in the contract documents and in ESS version 6. A copy of Relyant's second REA is attached hereto as Exhibit E.
- 19. Relyant supplied the additional labor and materials described in Relyant's two REAs in the course of its prosecution of the Subcontract work, at GOJV's request. Specifically, the additional work resulted from site conditions Relyant encountered in completing its scope of work under the Subcontract which differed materially from the contract documents and from the subsurface conditions Relyant reasonably expected to encounter in performing work on the Project.
- 20. Despite submitting its REAs more than two years ago, Relyant has received no payments from GOJV. After Relyant submitted its REAs, GOJV provided incomplete and inconsistent information to Relyant about the status of the two REAs.
- 21. Relyant has not been paid for any portion of the additional labor and materials provided under the Subcontract, as described in Relyant's two REAs. However, GOJV reached a settlement with the Government, without Relyant's knowledge or approval, to accept a

substantially reduced amount in full satisfaction of both of Relyant's REAs. GOJV's actions in this regard constituted a breach of its obligations under the Subcontract. Moreover, despite its settlement with the Government, GOJV has not paid Relyant even for the reduced amounts it received in connection with Relyant's REAs.

- 22. GOJV had an obligation under the Subcontract, including without limitation the express provisions in Sections 14.4 and 23.4 and the implied covenant of good faith and fair dealing, to reasonably prosecute the REAs with the Government on Relyant's behalf. GOJV breached the Subcontract by, *inter alia*, failing or refusing to take reasonable action to prosecute Relyant's REAs with the Government, by failing to make payment to Relyant of the amounts sought in the REAs, by mishandling the claims process with the Government involving Relyant's REAs, by accepting an unreasonable amount from the Government in settlement of Relyant's REAs, and by failing to keep Relyant reasonably informed about the status of its REAs.
- 23. Despite Relyant's repeated demands for payment, GOJV has failed to make payment for well over ninety (90) days to Relyant in the amount of \$2,682,962.13 (the sum of Relyant's unpaid Invoice # 20147-26 and the two REAs on the Project) for GOJV's breach of the Subcontract (including its breach of the implied covenant of good faith and fair dealing), which amount is and has been justly due and owing to Relyant.
- After a hearing held on February 17, 2025, an arbitrator appointed by the American Arbitration Association pursuant to the dispute resolution procedures described in the Subcontract entered her final award on March 19, 2025, in which the arbitrator determined that GOJV breached the Subcontract and awarded Relyant \$2,682,962.13, plus interest on that amount at 6%, as well as arbitration costs in the amount of \$19,675.00.

- 25. Under the Payment Bond, Defendants have bound themselves to make payment to claimants, such as Relyant, for all amounts due for materials and labor used in connection with the Project, including the amounts described in Relyant's unpaid Invoice # 20147-26 and two REAs.
- 26. As Relyant has complied with all requirements and conditions precedent applicable to it under the Subcontract and the Miller Act Payment Bond and has performed the work and provided the materials for which Relyant seeks payment, Relyant is entitled to payment of the amounts sought on the Project from Defendants on the bond issued for the Project.
- 27. This lawsuit is commenced within one year from the date upon which the last of the labor was performed and material and equipment was supplied by Relyant on the Project, as Relyant's scope of work on the Project is not yet complete, and Relyant's final labor on the Project has not yet been performed.

CLAIM ON PAYMENT BOND

- 28. Defendants undertook a direct obligation with the United States of America to act as the surety for the Project. In accordance with that obligation, Defendants issued a Miller Act payment bond securing the debts and obligations of GOJV, the prime contractor, for the Project.
- 29. Relyant performed all the work that it was contracted to perform and submitted REAs for the additional work that it performed on the Project pursuant to the Subcontract between GOJV and Relyant, which remain unpaid as of the date of the filing of this Complaint.
- 30. As a result of GOJV's breach of its payment obligations, Relyant has sustained damages for which it now seeks to recover. Pursuant to their obligations under the Miller Act, 40 U.S.C. §3131 *et. seq.*, Defendants, as co-sureties, are liable to Relyant for the unpaid amount of \$2,682,962.13 (the sum of Relyant's unpaid Invoice # 20147-26 and its two REAs on the Project) under the Payment Bond.

31. Based upon GOJV's failure to make timely payment to Relyant, Relyant is also entitled to recover from Defendants interest as allowed under the federal Prompt Payment Act, 31 U.S.C. § 3901 et seq.

WHEREFORE, Plaintiff Relyant Global, LLC prays for judgment in favor of Relyant Global, LLC against Defendants Travelers Casualty and Surety Company of America, The Continental Insurance Company, Federal Insurance Company, Zurich American Insurance Company, and Safeco Insurance Company of America, jointly and severally, in the unpaid amount of \$2,682,962.13 (the sum of Relyant's unpaid Invoice # 20147-26 and its two REAs on the Project) under the Payment Bond, as well as interest for the bond amount, at the rate allowed under the federal Prompt Payment Act, 31 U.S.C. § 3901 et *seq*.

Respectfully submitted this 18th day of July, 2025.

LAW OFFICE OF DANIEL J. BERMAN

Attorneys for Plaintiff RELYANT GLOBAL, LLC

By:

DANIEL J. BERMAN

Exhibit "A"

EXHIBIT A

Granite Bond Nos. 106770993 (TRA), 30008497 (CONT), 82450991 (FED)

Obayashi Bond Nos. 9263059 (FID/ZUR), 82	.458373 (FED), 070024672 (SAF),	929630027	(CONT)		
PAYMENT BOND (See instructions on reverse)	DATE BOND EXECUTED (Must be same or lat contract) August 18, 2017		OMB Control Number: 9000-0045 Expiration Date: 7/31/2019			
Paperwork Reduction Act Statement - This information collection 1995. You do not need to answer these questions unless we dis 9000-0045. We estimate that it will take 60 minutes to read the i suggestions for reducing this burden, or any other aspects of t Street, NW, Washington, DC 20405.	play a valid Office of Management and Budget (nstructions, gather the facts, and answer the que	DMB) control nun estions. Send anl	nber. The C	MB co relatin	ntrol number for t	his collection is mate, including
PRINCIPAL (Legal name and business address)		TYPE OF ORG	ANIZATION	("X" o	ne)	
Granite-Obayashi a Joint Venture		INDIVIDUAL PARTNERSHIP XJOINT				
585 West Beach Street Watsonville CA 95076		CORPORATION OTHER (Specify) STATE OF INCORPORATION				
		MAL.	N/A	4		
SURETY(IES) (Name(s) and business address(es))		PENAL SUM OF BOND				
		MILLION(S) \$164		ND(S) 893	HUNDRED(S) 061	CENTS 00
Please see sections A - F		CONTRACT DATE CONTRACT NUMBER		Fig. 18		
		August 17	7, 2017	1	N62742-17	-C-1324
			1 186			100

OBLIGATION:

We, the Principal and Surety(ies), are firmly bound to the United States of America (hereinafter called the Government) in the above penal sum. For payment of the penal sum, we bind ourselves, our heirs, executors, administrators, and successors, jointly and severally. However, where the Sureties are corporations acting as co-sureties, we, the Sureties, bind ourselves in such sum "jointly and severally" as well as "severally" only for the purpose of allowing a joint action or actions against any or all of us. For all other purposes, each Surety binds itself, jointly and severally with the Principal, for the payment of the sum shown opposite the name of the Surety. If no limit is indicated, the limit of liability is the full amount of the penal sum.

CONDITIONS:

The above obligation is void if the Principal promptly makes payment to all persons having a direct relationship with the Principal or a subcontractor of the Principal for furnishing labor, material or both in the prosecution of the work provided for in the contract identified above, and any authorized modifications of the contract that subsequently are made. Notice of those modifications to the Surety(ies) are waived.

WITNESS:

The Principal and Surety(ies) executed this payment bond and affixed their seals on the above date.

G	ranite-Oba	yashi a Joint Venture	PRINCIPA	AL				- Asset Heren Free Co.
SIGNATURE(S)		1. (Seal)	2.	3. (Seal) (Seal)		Corporate		
TIT	ME(S) & LE(S) ped)	yigisha Desai, Attorney-in-Fact	2.		3.			Seal
			INDIVIDUAL SUF	RETY(IES	3)			
SIG	SNATURE(S)	1.	(Seal)	2 (S			(Seal)	
	ME(S) ped)	1.		2. Some onlying to declarate the street				
			CORPORATE SU			West-filler in the		
4	NAME & ADDRESS	Travelers Casualty and Surety Company of America One Tower Square, Hartford CT 06183		STATE OF	INCORPORATION CT	LIABILITY LIMIT \$ 36,276,473.4	12	
SURETY	SIGNATURE(S)	1. Cu		Surety Phone No. 860, 277, 0111 1 1			Corporate Seal	
ns	NAME(S) & TITLE(S) (Typed)	1. C.K. Nakamura , Att	orney-in-Fact	2.			garanteeta - o - A	

AUTHORIZED FOR LOCAL REPRODUCTION Previous edition is NOT usable

STANDARD FORM 25A (REV. 8/2016) Prescribed by GSA-FAR (48 CFR) 53.2228(c)

	CORPORATE SURETY(IES) (Continued)			
NAME & ADDRESS	The Continental Insurance Company 333 S. Wabash Ave., Chicago IL 60604	STATE OF INCORPORATION PA	LIABILITY LIMIT \$ 42,047,730	56	
SIGNATURE(S)	1. Com	2. Surety Phone No. 312-822-5000			Corporate Seal
NAME(S) & TITLE(S) (Typed)	C.K. Nakamura , Attorney-in-Fact	2.			
NAME & ADDRESS	Federal Insurance Company 202B Hall's Mill Road, Whitehouse Station NJ 0888	STATE OF INCORPORATION LIABILITY LIMIT \$ 42,047,730.5		56	
SIGNATURE(S)	1 Oll	2. Surety Phone No. 908-903-2000			Corporate Seal
NAME(S) & TITLE(S) (Typed)	C.K. Nakamura , Attorney-in-Fact	2.			
NAME & ADDRESS	Fidelity and Deposit Company of Maryland 1299 Zurich Way, 5th Fl, Schaumburg IL 60196	STATE OF INCORPORATION MD	LIABILITY LIMIT \$ 14,000,000	.00	
SIGNATURE(S)	· CM	2. Surety Phone No. 847-605-6000			Corporate Seal
NAME(S) & TITLE(S) (Typed)	C.K. Nakamura , Attorney-in-Fact	2.			
NAME & ADDRESS	Zurich American Insurance Company 1299 Zurich Way, 5th FI, Schaumburg IL 60196	STATE OF INCORPORATION NY	LIABILITY LIMIT \$ 15,680,750	98	
SIGNATURE(S)	1. Ou-	2. Surety Phone No. 847-605-6000			Corporate Seal
NAME(S) & TITLE(S) (Typed)	1. C.K. Nakamura , Attorney-in-Fact	2.			
NAME & ADDRESS	Safeco Insurance Company of America 175 Berkeley St., Boston MA 02116	STATE OF INCORPORATION NH	LIABILITY LIMIT \$ 14,840,375	.48	
SIGNATURE(S	1. Oppor	2. Surety Phone No. 617-357-9500			Corporate Seal
NAME(S) & TITLE(S) (Typed)	1. C.K. Nakamura , Attorney-in-Fact	2.			
NAME & ADDRESS		STATE OF INCORPORATION	LIABILITY LIMIT		
SIGNATURE(S	1.	2.			Corporate Seal
SIGNATURE(S NAME(S) & TITLE(S) (Typed)	1.	2.			

INSTRUCTIONS

- 1. This form, for the protection of persons supplying labor and material, is used when a payment bond is required under 40 USC Chapter 31, Subchapter III, Bonds. Any deviation from this form will require the written approval of the Administrator of General Services.
- 2. Insert the full legal name and business address of the Principal in the space designated "Principal" on the face of the form. An authorized person shall sign the bond. Any person signing in a representative capacity (e.g., an attorney-in-fact) must furnish evidence of authority if that representative is not a member of the firm, partnership, or joint venture, or an officer of the corporation involved.
- 3. (a) Corporations executing the bond as sureties must appear on the Department of the Treasury's list of approved sureties and must act within the limitations listed therein. The value put into the LIABILITY LIMIT block is the penal sum (i.e., the face value) of the bond, unless a co-surety arrangement is proposed.
- (b) When multiple corporate sureties are involved, their names and addresses shall appear in the spaces (Surety A, Surety B, etc.) headed "CORPORATE SURETY(IES)." In the space designated "SURETY(IES)" on the face of the form, insert only the letter identifier corresponding to each of the sureties. Moreover, when co-surety arrangements exist, the parties may allocate their respective limitations of liability under the bonds, provided that the sum total of their liability equals 100% of the bond penal sum.
- (c) When individual sureties are involved, a completed Affidavit of Individual Surety (Standard Form 28) for each individual surety shall accompany the bond. The Government may require the surety to furnish additional substantiating information concerning its financial capability.
- 4. Corporations executing the bond shall affix their corporate seals. Individuals shall execute the bond opposite the words "Corporate Seal", and shall affix an adhesive seal if executed in Maine, New Hampshire, or any other jurisdiction requiring adhesive seals.
- 5. Type the name and title of each person signing this bond in the space provided.

STANDARD FORM 25A (REV. 8/2016) BACK

Exhibit "B"

EXHIBIT B

Reference No.: **733700** Subcontract No.: **08.0040**

Page 1 of 31

SUBCONTRACT

THIS SUBCONTRACT (hereafter "Subcontract"), is made and entered into this 1st day of October 2020 by and between GRANITE-OBAYASHI a Joint Venture (hereafter "Contractor") and Relyant Global, LLC located at 125 Tun Jesus Crisostomo Street, Tamuning, Guam 96913 (hereafter "Subcontractor").

The Contractor has entered into a contract with Naval Facilities and Engineering Command (NAVFAC) Pacific (hereafter "Client") for the performance of certain work more particularly described in such contract (hereafter "Contract") for work on the project (described below). The Contract includes but is not limited to, documents incorporated by reference or otherwise attached to the Contract, which may include the general, supplementary, special and other terms and conditions and the addenda, plans, drawings, maps and other documents. The Contract is expressly incorporated herein as a part of the Subcontract Documents (as defined in Article 1.1). The project (hereafter "Project") is generally described as follows: J-001B Finegayan Utilities and Site Improvements Phase I under contract number N62742-17-C-1324.

Contractor and Subcontractor desire to enter into this Subcontract whereby Subcontractor shall undertake the performance of a part of the work to be performed by Contractor under the Contract.

NOW, THEREFORE, in consideration of the mutual promises and conditions herein contained, Contractor and Subcontractor agree as follows:

- 1.0 SUBCONTRACT DOCUMENTS: 1.1 The Subcontract Documents shall consist of this Subcontract, all attachments to this Subcontract, all documents referenced or identified in the Subcontract and attachments to this Subcontract, and the Contract. The Subcontract Documents not physically attached to the Subcontract will, upon the written request of Subcontractor, be provided to Subcontractor or Contractor shall make such documents available for Subcontractor's review in Contractor's office during normal business hours.
- 1.2 Subcontractor shall be bound and obligated to Contractor by the terms and conditions of the Subcontract and the Subcontract Documents. Subcontractor assumes toward Contractor all obligations, liabilities and responsibilities that Contractor, by the Contract, has assumed toward Client as it applies to Subcontractor and/or the work of Subcontractor. Contractor shall further have the benefit of all rights, remedies, redress and limitations in respect to Subcontractor and all things done and used by Subcontractor in performance of its Work (as defined in Article 2.0) which Client and its agents have against Contractor under the Contract or by Law (as defined in Article 9.0). Any and all decisions or interpretations by Client or its agents relative to interpretation of the Contract or any ambiguity or discrepancy therein as it applies to the Work shall be binding on the Subcontractor to the same extent such decisions or interpretations are binding on Contractor.
- 1.3 The intent of the Subcontract Documents is to include all items necessary for the proper execution and completion of the Work (as defined in Article 2.0). The Subcontract Documents are complementary, and what is required by any one shall be as binding as if required by all with respect to the Work. Work not explicitly covered in the Subcontract Documents shall be performed by Subcontractor so long as it is consistent with and is reasonably inferable from the Subcontract Documents and shall be considered included in the Subcontract Price (as defined in Article 3.0). The Contract and this Subcontract shall be interpreted together and in harmony with one another. In the event of conflict between the Contract and the Subcontract, the Contract shall be controlling except with respect to the independent relationship between Contractor and Subcontractor, which shall be governed by the Subcontract.
- 1.4 Subcontractor hereby represents and acknowledges that it has reviewed and inspected all of the Subcontract Documents and has compared all such documents directly or indirectly relating to the performance of this Subcontract. Any and all errors, ambiguities and inconsistencies therein have been reported to the Contractor in writing and resolved to Subcontractor's satisfaction.
- **2.0 WORK TO BE PERFORMED: 2.1** Subcontractor, for the Subcontract Price (as defined in Article 3.0), shall furnish all supervision and labor, and furnish, supply and install all equipment, materials, and supplies and do all things necessary to fully, timely and properly perform and complete that portion of the work provided for in the Contract as identified in Attachment A.1 (hereafter "Work").
- 2.2 Subcontractor warrants to Contractor and Client that all Work shall be performed in a neat, skillful, good and workmanlike manner and shall be fit for its intended use both as to workmanship and materials. All materials and equipment furnished by Subcontractor for incorporation into the Project shall be new and of the best description and quality of their respective kinds, unless otherwise specified and ordered by Contractor in writing. Subcontractor warrants that the materials and equipment furnished and the Work performed will strictly comply with the Contract and this Subcontract and all applicable Laws (as defined in Article 9.0). Subcontractor acknowledges that Subcontractor's performance of the Work is subject to the approval and acceptance of Client to the extent required by the Contract and shall be reasonably satisfactory to Contractor.



Reference No.: 733700 Subcontract No.: 08.0040 Page 2 of 31

3.0 PAYMENT: In consideration of the promises, covenants and agreements of Subcontractor herein contained, and the full, proper and timely performance of the Work and the whole thereof in the manner and according to the requirements of the Subcontract Documents and instructions of Contractor and the satisfaction of all other requirements of Subcontractor contained in the Subcontract Documents, Contractor shall pay to Subcontractor and Subcontractor agrees to receive and accept as full compensation therefore, payment made on the basis of (i) lump sum; (ii) unit price(s); (iii) time and materials; and/or (iv) cost plus an established mark-up for overhead and profit (or any combination of payment terms i-iv) as set forth in Attachment A.1 (hereafter "Subcontract Price"). The Subcontract Price shall be subject to additions and deductions for changes in the Work as provided for in Article 14.0 and other adjustments as provided for in this Subcontract.

- **4.0 PAYMENT OF CHARGES AND TAXES**: The Subcontract Price includes, and Subcontractor shall promptly pay or cause to be paid, all costs, expenses, and other obligations incurred by, through or under Subcontractor related to the Project or the Work, including the cost of its sub-subcontractors of any tier or the suppliers to any of them and all taxes, including sales or use taxes, gross receipts taxes, assessment fees and charges, unemployment insurance, old age benefits, pensions or annuities imposed or required by any Law (as defined in Article 9.0) or government or labor agreement to which Contractor is a party or is otherwise obligated, and all other costs, expenses and obligations levied, incurred or payable on or in respect to Subcontractor's acts or transactions or tools, equipment or materials used by it in performing such obligations.
- **5.0 PROGRESS AND FINAL PAYMENTS: 5.1** Subcontractor, upon the request of Contractor, shall, prior to Subcontractor's first application for payment, submit to Contractor a schedule of values allocated to the various portions of the Work. The schedule of values shall reflect an accurate apportionment of the cost to perform the respective portions of the Work and shall be subject to the approval of Contractor and shall be in such form and supported by such data as Contractor may reasonably require. Subject to §5.3, the schedule of values shall be used for the evaluation of Subcontractor's application for payment.
- 5.2 Subcontractor, upon the request of Contractor, and on or before such date as Contractor shall designate, shall submit to Contractor, in form and content acceptable to Contractor, an itemized application for payment. The application for payment shall be notarized if required by Contractor and supported by such data substantiating Subcontractor's right to payment as Contractor may reasonably require including (i) invoices of sub-subcontractors or suppliers, (ii) certified payroll reports, (iii) reports of test results, (iv) inspection reports, or (v) any other documents required by this Subcontract or the Contract. If Subcontractor fails to timely submit its application for payment in form and content satisfactory to Contractor, a progress payment may not be included in Contractor's application for payment to Client.
- 5.3 Provided Subcontractor has complied with the requirements of §5.2, unless otherwise provided by Law (as defined in Article 9.0) or by the Subcontract Documents, within seven (7) business days after the date Contractor receives a progress payment from Client covering the amount of Work performed by Subcontractor, Contractor will pay to Subcontractor the amount due Subcontractor as estimated by Client (or Contractor, if Client makes no separate estimate of such Work, and approved by Client) for the period covered by the application for payment, less such amounts as Contractor shall determine as being properly withheld as allowed under this Subcontract. Subcontractor shall make payment to each of its subcontractors and suppliers amounts due them in the same manner and within the same time period as set forth above. Contractor shall reduce each payment to Subcontractor by the lesser of (i) zero percent (0%) or (ii) the maximum amount of retention allowed by Law (as defined in Article 9.0). To the fullest extent allowed by Law, the amount of reduction retained and withheld by Contractor under this §5.3, shall be released to Subcontractor only upon satisfaction of the requirements set forth in §5.6. Any sums that Contractor may receive from Client by reason of depositing security with Client shall be excluded in computing the payments to Subcontractor. The estimates of Client as to the amount of compensable Work performed by Subcontractor (or Contractor if Client makes no separate estimate of such Work) shall be binding upon Subcontractor and shall conclusively establish the amount of compensable Work performed by Subcontractor.
- 5.4 As a condition precedent to receiving payments from Contractor for Work performed, Subcontractor (and, to the extent required by Contractor, each of Subcontractor's subcontractors of any tier, the suppliers to any of them, and any other person or entity who has performed (or claims to have performed) a portion of the Work or a trust fund to which a portion of a laborer's total compensation is to be paid pursuant to an applicable employment agreement or collective bargaining agreement (collectively "Interest Claimants" and individually "Interest Claimant")) shall execute and deliver to Contractor, with its application for payment, a full and complete release of all claims (including, without limitation, any claim of lien or stop notice) and causes of action Subcontractor and such Interest Claimants, or any of them, may have (or claim to have) against Contractor, Client or the Project itself, any construction lender providing financing for the Project, or payment bond through the date of the execution of said release, save and except those claims specifically listed on said release and described in a manner sufficient for Contractor to identify such claim or claims with certainty. The release shall conform to the requirements of any applicable statute. All payments shall be deemed advances and are subject to adjustment at any time for errors, overpayment or Contractor's good faith determination that the remaining balance of payments may be insufficient to insure completion of the Subcontract Work in accordance with its terms or to pay lien, retention or bond claims.





Reference No.: **733700** Subcontract No.: **08.0040** Page 3 of 31

5.5 As a condition precedent to Contractor's obligation to make payments to Subcontractor for Work performed, Subcontractor shall have properly executed and delivered to Contractor this Subcontract and all documents and information to be furnished by Subcontractor as required by the Subcontract Documents including the certificate(s) of insurance and endorsements required under Article 10.0 and, if requested by Client or Contractor, the certification required under §9.2.

5.6 Unless otherwise provided by Law (as defined in Article 9.0) or by the Subcontract Documents, upon complete and satisfactory performance of this Subcontract by Subcontractor and final approval and acceptance of the Work by Client, Contractor will make final payment to Subcontractor of the balance due under this Subcontract within the lesser of (i) seven (7) business days after receipt by Contractor of final payment from Client for such Work or (ii) such time as prescribed by Law or the Subcontract Documents provided, however, that Subcontractor, following the request of Contractor, has timely furnished to Contractor (1) an affidavit that all payrolls, taxes, bills for material and equipment and other indebtedness connected with the Work for which Client, Client's property or Contractor might in any way be responsible, have been paid or otherwise satisfied; (2) consent of surety, if any, to final payment; (3) other data establishing payment or satisfaction of all such obligations to the extent and in such form as may be required by Client and/or Contractor; (4) fully executed waivers and/or release of (i) liens, (ii) notices to withhold or (iii) other claims against the land, improvements, or funds involved in the Project from Subcontractor and from Interest Claimants in form satisfactory to Contractor or required by statute, as appropriate; and (5) all other documents required by Client or the Subcontract Documents as a condition to final payment. If any Interest Claimant or other person or entity refuses to furnish a release or waiver or any other document or data required by Client or Contractor or as otherwise required under this §5.6, the Subcontractor shall, upon the request of Contractor, furnish security of a nature and in a form and amount required by Contractor including a bond in an amount prescribed by statute (or if not prescribed by statute in an amount as reasonably determined by Contractor) and form reasonably satisfactory to Client and/or Contractor which security shall be sufficient to protect and hold harmless Client and Contractor against any such claim or lien filed or presented by an Interest Claimant. Should the existence of any unsatisfied or undischarged claim, obligation or lien arising in conjunction with the Work become known after final payment is received from Contractor, Subcontractor shall promptly pay on demand all actual amounts Contractor and/or Client pay in putting in place and maintaining a release bond conforming to statutory requirements if applicable, or in satisfying, discharging or defending any such claim, obligation or lien, including all costs and actual attorney's fees incurred in connection therewith. Notwithstanding the payment to Subcontractor of final payment in accordance with §5.6 (and the satisfaction of any conditions required as a pre-requisite to the payment by Contractor to Subcontractor of final payment under §5.6 or any other Subcontract Document), Subcontractor shall continue to be obligated to Contractor under this Subcontract, including Article 14, until such time as Contractor is relieved of any further obligations to Client under the Contract.

5.7 Contractor, to the extent allowed by Law (as defined in Article 9.0), may withhold from Subcontractor payments otherwise due, in whole or in part, or may nullify, in whole or in part, any approval previously made by Client or Contractor on account of defective materials or defective Work not remedied; claims filed or reasonable evidence indicating probability of filing of claims (including a claim of lien or stop notice) by an Interest Claimant; failure of Subcontractor to timely make the required payments to its subcontractors or others for material, supplies or labor or government agencies owed taxes; incomplete or non-conforming application for payment (if required); Subcontractor's failure to carry out or perform the Work in accordance with the Subcontract Documents; or any other reason set forth in the Subcontract Documents pertaining to Client's withholding of payments to Contractor. If the foregoing causes are promptly removed or corrected to Contractor's satisfaction, the withheld payments shall be made. If such causes are not promptly removed or corrected, Contractor may rectify the same at Subcontractor's expense including reimbursement of Contractor's actual attorney fees and costs incurred and the cost of procuring a release bond necessary to remove the effect of a claim of lien or stop notice where a lien or stop notice has been filed. No certificates issued or payment made to Subcontractor shall be considered an acceptance of any Work, whether or not defective or improper. Should any claim against Contractor arise after final payment is made, all provisions of Article 11.0 shall apply and be binding on Contractor and Subcontractor.

5.8 Subcontractor grants to Contractor the right of set off of sums due to Subcontractor under the provisions of this Subcontract against any obligation that may be due from Subcontractor or its affiliates to Contractor for this Subcontract. Subcontractor, upon the request of Contractor, shall furnish an affidavit showing the names and addresses of all government agencies owed taxes or persons or companies that will or have furnished labor, materials, equipment or services for the Work and the amount due or to become due to each such person or company. No portion of the Subcontract Price otherwise payable or to be paid to Subcontractor shall be considered actually earned by Subcontractor until all such persons or companies are fully paid by Subcontractor in accordance with applicable Subcontract terms and conditions.

5.9 To the fullest extent permitted by Law (as defined in Article 9.0), Contractor's payment to Subcontractor shall be conditioned upon Contractor's receipt of payment from Client for Work performed by Subcontractor. If Client has not timely paid Contractor all or any portion of amounts due Contractor for Work performed by Subcontractor, and Contractor has initiated efforts to obtain payment from Client (including, but not limited to, litigation, arbitration or other form of alternate dispute resolution), then Contractor's obligation to pay Subcontractor shall be deferred until the first to occur of (i) ten (10) business days following receipt of payment by Contractor from Client; or





Reference No.: **733700**Subcontract No.: **08.0040**Page 4 of 31

- (ii) twenty (20) business days following the conclusion or failure of such effort. It is expressly agreed that such deferral is a reasonable period of time; provided, however, if, during such deferral period, the statute of limitations for filing an action to foreclose a lien is set to expire, then no more than twenty (20) business days prior to the expiration of such statute of limitations, Subcontractor may take the action required in order to preserve its lien rights; and provided further, in no event shall Subcontractor file any action to foreclose its lien rights prior to completion of the work of improvement.
- 5.10 Subcontractor agrees and covenants that money received for the performance of this Subcontract shall be held in trust and used solely for the benefit of employees of Subcontractor and firms supplying labor, materials, supplies, tools, machines, equipment, plant or services for this Project in connection with this Subcontract and having the right to assert liens or other claims against the land, improvements or funds involved in this Project or against any bond or other security posted by Contractor or Client.
- 5.11 Subcontractor warrants that title to all Work (including materials and equipment) covered by an application for payment will pass to Client, free and clear of all liens, claims, security interests or encumbrances upon the first to occur of (i) incorporation into the Project or (ii) upon the receipt of payment therefore by Subcontractor and that no Work, materials or equipment covered by any application for payment will have been acquired by the Subcontractor, or by any other person performing the Work at the site or furnishing materials and equipment for the Project, subject to an agreement under which an interest or an encumbrance is retained by the seller or otherwise imposed by the Subcontractor or such other person.
- 5.12 To the extent allowed by Law (as defined in Article 9.0), payments, including progress and retention payments, may, in the discretion of Contractor, be made in the form of checks payable jointly to an Interest Claimant. Subcontractor shall immediately pay all Interest Claimants all amounts they are due. If Subcontractor shall fail to timely pay when due an Interest Claimant for any taxes, materials, labor and services furnished in connection with the performance of the Work, Contractor may pay in its sole discretion (but without obligation to do so) the amount of such taxes, labor, services and/or materials directly to the entity or person providing the same or the governmental agency to whom taxes are due. The amount paid by Contractor shall be deducted from payments otherwise due, or to become due, Subcontractor. Such payments shall be deemed payments made by Contractor to Subcontractor and the Subcontractor shall have no recourse against Contractor for payments made by Contractor hereunder.
- **5.13** Subcontractor shall not be entitled to any interest on retention or other amounts to be paid to Subcontractor under the terms of this Subcontract except where the payment of interest is required by statute and the right to such interest cannot be waived or where Contractor receives interest from Client on proceeds due Subcontractor for Work performed.
- 6.0 WORK AND JOBSITE CONDITIONS; REVIEW OF EARLIER WORK: 6.1 Subcontractor warrants and represents that it has inspected and familiarized itself with the site of its Work and the conditions affecting the Work including those bearing on transportation, disposal, handling and storage of materials, availability of labor, utilities, roads and uncertainties of weather, river stages, tides or other physical conditions of the site and the type of equipment and facilities needed preliminary to and during the performance of the Work. Subcontractor has satisfied itself as to the surface and subsurface conditions of the site from inspection of the site, all exploratory work done by Client and/or Contractor, as well as from information presented in the Subcontract Documents. Any failure by Subcontractor to acquaint itself with such available information that results in Subcontractor reaching erroneous conclusions or misinterpretations shall preclude Subcontractor from being entitled to a change in Subcontract Price or time extension. Subject to Subcontractor's compliance with the requirements of Article 14.0 and the Subcontract Documents, Subcontractor shall be entitled to an equitable proportionate share of any compensation Contractor receives from Client for changed or differing site conditions which impact Subcontractor's performance of the Work but only to the extent received by Contractor.
- **6.2** Before proceeding with its work under this Subcontract, Subcontractor will verify that all earlier work is ready to accept the Work. By proceeding with the Work, Subcontractor accepts all earlier work as satisfactory and will pay resulting expenses if the earlier work is not fully satisfactory, including the cost to remove and replace any such work.
- **7.0 INDEPENDENT CONTRACTOR: 7.1** Subcontractor represents and warrants that it is an independent contractor and shall operate as an independent contractor and not as an agent of Contractor. Subcontractor shall not represent or hold itself out to be an agent of Contractor. This Subcontract does not create, nor does any course of conduct between Contractor and Subcontractor with respect to this Subcontract create, any contractual relationship between any parties other than as Contractor and Subcontractor. Subcontractor shall assume all of the rights, obligations and liabilities, applicable to it as an independent contractor hereunder. Any provisions in this Subcontract which may appear or purport to give Contractor the right to direct Subcontractor as to details of performing the Work or to exercise a measure of control over the Work shall be deemed to mean that Subcontractor shall follow the Subcontract Documents to achieve the results of the Work only.
- 7.2 Subcontractor represents and warrants that it (i) is fully experienced and properly qualified to perform the class of Work provided for herein, (ii) has the financial capability and shall finance its own operations required for the performance of the Work and (iii) is properly equipped and organized to perform the Work in a competent, timely and proper manner in accordance with the requirements of the





Reference No.: **733700** Subcontract No.: **08.0040** Page 5 of 31

Subcontract Documents. Subcontractor represents and warrants that its entry into this Subcontract has not been induced either wholly or in part by any promises, representations or statements by Contractor or its agents or representatives other than those representations set forth in this Subcontract and that this Subcontract fully integrates the intentions of Contractor and Subcontractor. Subcontractor represents and warrants that is not presently debarred, suspended, proposed for debarment, or declared ineligible for the award of contracts by any federal or other public agency.

- **8.0 COMMENCEMENT AND PROGRESS OF WORK: 8.1** Time is of the essence in the performance by Subcontractor of its obligation under this Subcontract. Subcontractor shall initiate the Work and shall thereafter proceed and complete performance of the Work promptly and diligently all in accordance with the Work Progress Schedule (as defined in §8.3). The Work shall be performed in such a manner and sequence as to assure the timely completion of other work dependent on the Work and to permit completion of the Work as required by the Work Progress Schedule (as defined in §8.3). The timely performance of the Work in accordance with the Work Progress Schedule (as defined in §8.3) is a material provision of the Subcontract.
- **8.2** Except as otherwise directed by Contractor, Subcontractor shall, within five (5) business days after its execution of this Subcontract, or within such other time as may be specified by Contractor, prepare and submit to Contractor in writing for Contractor's review and approval a proposed schedule, showing in detail (i) the number of days required by Subcontractor to complete the Work; (ii) the order in which Subcontractor proposes to carry out all major activities (including the sequence and anticipated duration of such activities) anticipated by Subcontractor and (iii) the dates on which the activities representing the complete performance of the Work (including procurement of materials, plant and equipment) may be started and finished ("Proposed Schedule"). Where Contractor has made available to Subcontractor a schedule for the Project as approved by Client ("Client's Schedule"), Subcontractor shall schedule and sequence its Work such that it is integrated into and consistent with the Client's Schedule.
- 8.3 Provided Contractor has timely received the Subcontractor's Proposed Schedule for performance of the Work (if required), Contractor shall prepare the work progress schedule incorporating Subcontractor's Proposed Schedule to the extent reasonably feasible and consistent with the Client's Schedule (as integrated "the Work Progress Schedule"). Should any aspect of the Work Progress Schedule be at variance with the Subcontractor's Proposed Schedule or desired sequencing or duration of its Work, Subcontractor shall notify Contractor in writing of such variation within five (5) business days of its receipt of the Work Progress Schedule. Thereafter, the parties shall meet in a good faith effort to resolve any scheduling differences. However, Subcontractor shall abide by a final determination made by Contractor, in Contractor's sole discretion, with respect to the scheduling of the Work including the sequencing and duration of the Work. In any event, Contractor reserves the right to require Subcontractor, upon five (5) business days advance written notice, to commence the Work. Subcontractor's failure to submit its Proposed Schedule, if and when required, or to notify Contractor in writing of any variation or conflict between Subcontractor's Proposed Schedule and the Work Progress Schedule shall bar Subcontractor from claiming either extensions of the time and/or activity completion dates specified in the Work Progress Schedule or an increase in the Subcontractor's inability to comply with the Work Progress Schedule.
- 8.4 Subcontractor further acknowledges that, as construction progresses, it may be necessary for Contractor to change the sequential order and duration of various activities, including those contemplated by the Subcontract to reflect unanticipated delays, occurrences and other factors which act to alter the Work Progress Schedule. Contractor may require Subcontractor to prosecute the Work in such sequence as Contractor may reasonably require in order to maintain the progress of others and of the Work Progress Schedule. Contractor may, from time to time, revise and update the Work Progress Schedule so as to reflect the actual progress of the Work and/or changes in the Work. Subcontractor shall coordinate its Work with that of Contractor and others as Contractor may reasonably require. Subcontractor in the reasonable opinion of Contractor, shall at all times furnish and have available such sufficient and satisfactory equipment, materials, supplies and labor to perform the Work to meet the Work Progress Schedule. In the event Subcontractor fails to perform the Work in accordance with the Work Progress Schedule or, in order to avoid falling behind the Work Progress Schedule, to the extent such failure is not the result of a delay for which Subcontractor is entitled to an extension of time in which to perform the Work under this Article 8.0, Subcontractor, at its own expense, shall promptly provide additional equipment and/or labor and work additional shifts or overtime so as to meet and maintain the Work Progress Schedule. Subcontractor shall be responsible for all expenses and damages incurred by Contractor resulting from the failure of the Subcontractor to meet the Work Progress Schedule.
- **8.5** If Subcontractor shall be delayed in the performance of the Work by any act or neglect of Client, its agents (including consultants, architects, engineers or other design professionals) or by changes ordered by Client in or to the Work, or other events beyond the control of Subcontractor and not due, in whole or in part, to the fault of Subcontractor, Subcontractor shall be entitled to an extension of time and/or compensation for additional costs incurred by Subcontractor as a result thereof, but only to the extent that an extension of time and/or compensation (including compensation for interferences, hindrances, inefficiencies or disruptions) is received by Contractor from Client for delays and/or costs incurred by Subcontractor and only if Subcontractor shall have notified Contractor in writing of the existence of the delay within two (2) business days of the occurrence of the event giving rise to the claim of delay. Subcontractor shall have the burden of proving the impact of the delay and justifying the time extension and/or compensation requested. If the performance of the Work by Subcontractor is





Reference No.: **733700** Subcontract No.: **08.0040** Page 6 of 31

unaffected by a delay incurred by Contractor (or another subcontractor) and for which relief is granted to Contractor (or such other subcontractor), Subcontractor shall not be entitled to claim an extension of time for such delay as a justification for delay in the timely performance of its Work.

9.0 LAWS AND REGULATIONS: 9.1 Subcontractor, its employees and representatives shall at all times comply with all applicable laws, ordinances, statutes, rules and regulations, federal, state, county and municipal, and rulings or directives of any agencies having jurisdiction including those relating to the environment, wages (including, if required, the payment of prevailing wages and/or the timely submission of certified payrolls), hours, safety, equal employment opportunity and working conditions or which pertain in any way to the scope of work to be performed by Subcontractor (herein "Laws" or "Law"). Subcontractor represents and warrants that it holds all licenses required by Law and necessary for Subcontractor to perform the Work. Subcontractor shall procure and pay for all licenses, inspections and permits required by any governmental authority or any Law for any part of the Work. For the purposes of this Article 9.0, the territory of Guam shall be deemed a "state."

9.2 If Subcontractor claims status as a disadvantaged business enterprise, including, without limitation, a minority business enterprise, veteran or service disabled veteran business enterprise, women's business enterprise, disabled persons business enterprise, HUB Zone small business, or any other specialized status or protected class recognized by applicable Law (herein "Specialized Status), then Subcontractor shall, at the request of Contractor (or Client) make all records available (including those of any lower tier subcontractor or supplier) in support of Subcontractor's claim (or that of a lower tier subcontractor or supplier) of Specialized Status within two (2) business days of such request to Contractor (or Client). Subcontractor shall provide to Contractor certification or other proof of Specialized Status (including confirmation of the Specialized Status claimed) (i) as of execution of this Subcontract; (ii) as of Subcontractor's application for final payment; (iii) within two (2) business days of a separate request made by Contractor therefore; and (iv) at such other times as Contractor may require. The certification or such other proof of status shall be in such form and content as required by Client and/or Contractor and shall include a statement certifying that Subcontractor is performing a commercially useful function as defined by applicable Law including 49 C.F.R. 26.55(c)(1). If the Specialized Status claimed by Subcontractor is denied or revoked by any agency or entity having authority to do so, then Subcontractor shall immediately provide written notice to Contractor of such loss of status together with the written determination of such agency or entity as communicated to Subcontractor.

10.0 INSURANCE: 10.1 Upon execution of this Subcontract and, in any event, before commencing the Work, Subcontractor shall procure, and maintain in full force and effect at Subcontractor's expense, so long as may be necessary to fully protect Client and Contractor and, in any event, at least through the end of the warranty period, including any special warranties, applicable to the Work, Workers' Compensation Insurance in accordance with the Laws of Guam; and Employers' Liability Insurance; Commercial General Liability Insurance on an occurrence basis only with the general aggregate applying on a per project basis and including products, completed operations, contractual liability, broad form property damage, coverage for explosion, collapse, underground damages, and independent contractors; Comprehensive Automobile Liability Insurance on an "Any Auto" basis covering any and all owned, non-owned and hired automobiles used in connection with the Work and such other coverage and in the minimum limits as specified in Attachment A.3. If higher limits or other forms of insurance are required under the Contract, Subcontractor will comply with such requirements. The insurance required to be provided under this Article 10.0 shall be issued by companies with a rating by A. M. Best of A- or better and a financial size classification of VIII or better.

10.1.1 If the Subcontractor's scope of work includes professional services, including design or engineering, the design professional shall procure and maintain Professional Errors and Omissions coverage written on a standard industry policy with minimum limits stated in Attachment A.3. The policy shall be subject to the approval of Contractor and, upon the request of Contractor, Subcontractor shall produce such policy for Contractor's review. Subcontractor shall provide Contractor with a certificate of insurance evidencing design professional's Professional Errors and Omissions coverage in accordance with the requirements of Attachment A.3.

10.2 In addition to the insurance coverages identified and required in §10.1, if Subcontractor or any of its subcontractors of any tier are required to perform abatement, removal, remediation, transportation or disposal of hazardous, toxic or contaminated material including asbestos, heavy metals, etc., ("HTC Material") Subcontractor must procure and maintain Contractor's Pollution Liability insurance coverage or its equivalent. Such coverage shall carry limits of not less than the limits identified in Attachment A.3 and shall conform to the requirements of this Article 10 including the additional insured obligation stated in §10.3. If the Contractor's Pollution Liability coverage is written on a "claims made" policy, it shall have at least a three (3) year extended reporting period beyond completion of the Work. Coverage shall include an MCS-90 endorsement to Subcontractor's Comprehensive Automobile Liability policy or an endorsement to the Contractor's Pollution Liability to extend to transportation and hauling of HTC Material.

10.3 The Commercial General Liability insurance policy (and the Contractor's Pollution Liability policy if the Work includes abatement, removal, remediation, transportation or disposal of HTC Material), Comprehensive Automobile Liability insurance policy and any and all excess policies maintained by Subcontractor shall be endorsed (utilizing the form of endorsement identified in **Attachment A.3**) to stipulate that the insurance afforded the additional insureds (as identified in the next sentence), including completed operations, shall apply as primary insurance without qualification and that any insurance maintained by Client and Contractor shall be in excess only and shall not be called





Reference No.: **733700**Subcontract No.: **08.0040**Page 7 of 31

upon to contribute with the insurance of Subcontractor. Contractor, Client and their directors, officers, employees and agents and such other persons or entities as Contractor may require or is required of Contractor, under the Contract Documents, shall be named, by endorsement (using a form of endorsement approved by Contractor), as additional insureds on Subcontractor's general liability insurance policy and its automobile liability insurance policy, and, if Subcontractor's scope of Work includes the abatement, removal, remediation, transportation or disposal of any HTC Materials, its Contractor's pollution liability insurance in accordance with the requirements set forth in **Attachment A.3**.

- 10.4 Before commencing the Work or within ten (10) business days after execution of this Subcontract whichever shall first occur, the Subcontractor shall furnish a certificate of insurance and endorsements (including the additional insured endorsement required under §10.3), satisfactory to Contractor, from each insurance company providing coverage to Subcontractor showing that said insurance coverage is in force, stating policy numbers, dates of expiration, deductible amounts applicable to each policy, and limits of liability thereunder, and further providing that the insurance will not be canceled, modified or changed until the expiration of at least thirty (30) calendar days after written notice of such cancellation, modification or change has been mailed by certified mail return receipt requested to Contractor. Subcontractor, upon the request of Contractor shall thereafter provide a current certificate of insurance and endorsements within ten (10) business days of the request of Contractor. Subcontractor shall be responsible for all deductible amounts applicable to each policy.
- 10.5 Subcontractor's insurance shall apply separately to each insured against whom claim is made or suit is brought, except with respect to the limits of the insurer's liability.
- 10.6 Subcontractor for itself and on behalf of its insurers hereby waives and releases the additional insureds from liability for loss, damage or loss of property at the job site, which loss is covered by said insurance. Subcontractor shall obtain a waiver of any subrogation right that its insurers may acquire against any of the additional insureds by virtue of payment of any such loss covered by such insurance.
- 10.7 If Subcontractor's existing coverage does not meet the requirements set forth herein, Subcontractor agrees to amend, supplement, or endorse the existing coverage to do so at no additional cost to Contractor. Compliance in full with theinsurance requirements stated in this Article 10.0, are understood to be a condition precedent to Contractor's obligations under this Subcontract.
- 10.8 The Subcontractor shall be responsible for determining the adequacy of insurance coverage requested, or put in place by others, with respect to the Work to be completed under this agreement. The Subcontractor shall, at its own expense, purchase any additional insurance which is required by Law or which it considers necessary to protect itself against liability or damage, loss or expense arising from damage to property or injury or death of any persons arising in any way out of, in conjunction with, or resulting from performance of the Work.
- 10.9 It is understood and agreed that authorization is hereby granted to the Contractor to withhold payments to the Subcontractor until a current and properly executed Certificate of Insurance and approved endorsements providing insurance as required herein is received by the Contractor.
- 11.0 LIABILITY FOR INJURY AND DAMAGES: 11.1 To the fullest extent permitted by Law, Subcontractor shall indemnify, defend (with counsel reasonably satisfactory to Contractor), and hold harmless Client and Contractor, as well as any individual and/or entity that Contractor is required by contract, including the Subcontract Documents, to indemnify, defend and/or hold harmless, and their officers, directors, joint venturers, agents, servants and employees, and each of them, (hereafter collectively "Indemnified Parties" and individually an "Indemnified Party"), from and against, to the proportionate extent, any and all, suits, actions, legal or administrative proceedings, claims, debts, demands, damages, consequential damages, liabilities, interest, reasonable attorney's fees, costs and expenses of whatever kind or nature (hereafter "Claim" or "Claims") and whether they may arise during or after performance of the Work which are in any manner directly or indirectly caused, occasioned or contributed to, in whole or in part, or claimed to be caused, occasioned, or contributed to, in whole or in part, through any act, omission, fault or negligence whether active or passive of Subcontractor, or anyone acting under its direction, control, or on its behalf or for which it is legally responsible, in connection with or incident to the Work or arising out of any failure of Subcontractor to perform any of the terms and conditions of this Subcontract; without limiting the generality of the foregoing, the same shall include injury or death to any person or persons (including Subcontractor's employees) and damage to any property, regardless of where located, including the property of Client and Contractor. Subcontractor, however, shall be relieved of and shall have no further obligation to indemnify an Indemnified Party under any construction contract upon final resolution of a Claim (i.e. a Claim from which there is no longer any right of appeal) to the extent such Claim is determined to be due to (i) negligence or willful misconduct of an Indemnified Party or such Indemnified Party's agents or employees, (ii) defects in design furnished by an Indemnified Party or (iii) to the extent the Claim does not arise out of, pertain to or relate to the Work.

Contractor agrees to indemnify and hold harmless the Subcontractor, its officers, directors, and employees against all damages, liabilities, or cost, including reasonable attorney's fees and defense costs, to the extent caused by the Contractor's negligence.

11.2 To the fullest extent permitted by Law, Subcontractor shall indemnify and defend and hold harmless the Indemnified Parties, and each of them, from and against all claims, royalties, damages, liabilities, costs and expenses of whatever kind or nature (including reasonable





Reference No.: **733700** Subcontract No.: **08.0040** Page 8 of 31

attorney's fees) in any manner resulting, or claimed to result from any alleged infringement of any letters patent or patent rights by reason of the Work or materials or processes used by Subcontractor, its lower tier subcontractors and/or suppliers or others acting on its behalf, in the performance of the Work.

- 11.3 The indemnification obligations of Subcontractor under this Subcontract including Articles 11.0, 12.0 and 25.0 and §5.6 shall survive termination of this Subcontract and shall extend to claims occurring after termination of this Subcontract as well as while it is in force.
- 11.4 The obligations for indemnification herein required are severable. In addition to the provisions of Article 26.0, the unenforceability of any portion of the obligation for indemnification hereunder due to the effect of any Law, court decision, or any other reason, shall not nullify, reduce or limit other obligations set forth herein and such remaining obligations shall be valid and enforceable to the fullest extent permitted by Law. All other obligations arising under this Article 11.0 shall remain in full force and effect.
- 11.5 The insurance obligations of the Subcontractor are separate, independent obligations under the Subcontract Documents, and the obligations for defense and indemnity contained in Articles 11.0, 12.0, and 25.0 and §5.6 are not intended to modify, nor should they be construed as modifying, or in any way limiting, the insurance obligations set forth in the Subcontract Documents.
- 11.6 Contractor shall be entitled to recover actual attorney fees and court costs and all other costs, expenses and liabilities incurred by Contractor in an action brought to enforce all or any part of this Article 11.0.
- 11.7 In the event any dispute arises over Subcontractor's duty to defend or indemnify an Indemnified Party (including, without limitation, any dispute relating to the responsibility of the parties with respect to the defense obligation or the scope of Subcontractor's duty to defend or indemnify as required herein, or Subcontractor's performance or non-performance under this Article 11.0 including the reallocation of defense costs), Contractor and Subcontractor agree, in addition to the rights and responsibilities of the parties under Article 23.0, to submit such dispute to binding arbitration before the American Arbitration Association, under the Construction Industry Arbitration Rules and Mediation Procedures, Fast Track Procedures.
- 11.10 Subcontractor shall ensure that its sub-subcontractors also fully indemnify, defend and hold harmless the Indemnified Parties to the same extent that Contractor is required to indemnify, defend and hold harmless such Indemnified Parties.
- 12.0 DELAY DAMAGES: In the event any acts or omissions by or of Subcontractor delay Contractor in the performance of the Work or any others performing work for the Project or on the site of the Project resulting in Contractor being subjected to any damage or liability whether under the Contract or through any third party action or proceeding, Subcontractor agrees to indemnify, defend and hold Contractor harmless against any damage or liability for and to promptly pay to Contractor any damage or liability so incurred (including liquidated or actual damages) and such other or additional damages, costs, assessments and expenses including actual attorneys' fees, as Contractor may sustain by reason of Subcontractor's delay or failure to perform.
- 13.0 ASSIGNMENT: 13.1 Subcontractor shall not assign, sublet, subcontract or hypothecate this Subcontract or any of the Work or any amount due or payable hereunder without the prior written consent of Contractor. Any assignment, subletting, subcontracting or hypothecating of this Subcontract without such consent shall be void.
- 13.2 Contractor has the right to assign this Subcontract to Client (or if Client is not the owner of the Project, to the entity designated by Client) consistent with the terms and conditions of the Contract. Subcontractor hereby agrees to accept such an assignment under the terms and conditions set forth in the Contract.
- 14.0 CHANGES AND EXTRA WORK: 14.1 Without invalidating this Subcontract and without notice to any surety, Contractor may, at any time or from time to time, order additions, deletions or revisions in the Work to be performed and/or the Work Progress Schedule ("Change" or "Changes"). Such changes may include any kind of addition, deletion, or modification of scope, for any purpose, and regardless whether added work changes the existing design or adds an entirely new element to the Work. Under no conditions shall Subcontractor make any changes to the Work, either as additions or deductions, without the written order of Contractor ("Change Order").
- 14.2 All Changes ordered by Contractor but which originate with or are requested by Client will be compensated for by Contractor but only to the extent a commensurate adjustment in compensation or time is allowed Contractor by Client for labor, materials, equipment and Markup (as defined in §14.5) for Subcontractor performing the Change and any impact that Change has on the Work.
- 14.3 All Changes directed by Contractor but which do not originate with or are not requested by Client will be compensated for by Contractor as mutually agreed in writing by Subcontractor and Contractor but in no event greater than Subcontractor's actual costs, plus Mark-up (as defined in §14.5) for performing the Change and any impact that Change has on the Work plus Mark-up (as defined in §14.5).





Reference No.: 733700 Subcontract No.: 08.0040 Page 9 of 31

14.4 In the event Subcontractor encounters any condition or event (other than a condition or event described in §8.5 or a condition or event caused solely by Contractor) during the performance of the Work that Subcontractor believes entitles it to an adjustment in the Subcontract Price and/or the Work Progress Schedule (a "Site Condition"), Subcontractor shall, before such Site Condition is disturbed, give written notice to Contractor of such Site Condition. The written notice shall be given to Contractor sufficiently in advance of the time Contractor is required to submit a claim, potential claim and notice of claim ("Potential Claim") to Client in accordance with the Contract so as to allow Contractor adequate time to understand and familiarize itself with the Site Condition, the impact that the Site Condition has or may have on Contractor and others, and the opportunity to prepare and timely submit to Client a Potential Claim on behalf of Subcontractor. The written notice shall have adequate information to enable Contractor to submit a Potential Claim on behalf of Subcontractor that is in strict accordance with the Contract. In no event, however, shall such written notice be given to Contractor more than two (2) business days after Subcontractor encounters the Site Condition. The timely delivery of an adequate written notice shall be an express condition precedent to the Subcontractor's right to an adjustment in the Subcontract Price and/or Work Progress Schedule because of such Site Condition. If Subcontractor fails to timely submit an adequate written notice in strict accordance with the requirements of this Subcontract, the Subcontractor waives any and all rights to any adjustment in the Subcontract Price and/or the Work Progress Schedule. In the event Contractor is allowed an adjustment in compensation and/or a time extension by Client as a result of a Site Condition for which Subcontractor provided a timely and adequate written notice, then Subcontractor shall be entitled to an adjustment in compensation to the extent a commensurate adjustment in compensation or time is allowed Contractor by Client for costs incurred by the Subcontractor that were caused by the Site Condition.

14.5 "Mark-up" is defined to be the overhead and profit that is specified in the Contract or any applicable Laws. However, if not so specified, Subcontractor's Mark-up shall be defined to be not more than 15% of the direct cost of the Change Order Work or an amount agreed to or established by Client or in the Subcontract Documents. Contractor shall be entitled to its own Mark-up from Client in addition to the Mark-up passed-through to Subcontractor for all Work performed under the Subcontract to the extent permitted by the Contract or by any applicable Laws or as agreed to by Client. Subcontractor agrees that payment of its Change Order fully compensates Subcontractor for all jobsite and home office overhead incurred by Subcontractor and its lower tier subcontractors and suppliers for the time period during which the work is performed under the Change Order.

14.6 Subject to the limitations stated in this Article 14.0, if pursuant to this Article 14.0 there is any Change or Subcontractor has encountered a Site Condition and has complied with all notice requirements in §14.4 and Contractor and Subcontractor are unable to agree on an adjustment to the Subcontract Price or method for adjusting the Subcontract Price (including unit price or force account rates) and/or adjustment in the Work Progress Schedule; or if a dispute arises between Contractor and Subcontractor as to (i) whether work performed by Subcontractor is included as part of its scope of Work; (ii) the validity of Contractor withholding any sums or exercising any of its rights under this Subcontract; or (iii) whether Subcontractor is entitled to additional time (including the amount of additional time) or compensation for the delays referred to in §8.5 or elsewhere in this Subcontract (collectively "Sub Claim"), then Subcontractor shall timely perform such work so as not to delay the Work Progress Schedule and the Sub Claim shall be resolved pursuant to Article 23.0.

14.7 Changes made under this Article 14.0 shall not be construed as interference with Subcontractor's performance of the Work regardless of the extent or number of such changes and whether the changes originate with the Contractor or Client including Client's or Contractor's exercise of the right to suspend all or any portion of the Work.

15.0 TERMINATION OR SUSPENSION OF WORK FOR CONVENIENCE: 15.1 If Client for any reason (including Client's convenience) terminates or suspends performance of the Work, or any portion of it under the Contract, Contractor may order Subcontractor to terminate or suspend its performance of the Work, or any portion of it and Contractor shall not be liable in any way for any such termination or suspension. Provided, however, to the extent that Contractor is compensated by Client by reason of such termination or suspension, then in such event Contractor shall pay to Subcontractor an equitable portion of said sum based upon the Work performed. In no event, absent Contractor's written determination to terminate this Subcontract, shall any change, directive or other communication from Client effect a termination or suspension of the Work or entitle Subcontractor or any of its sub-subcontractor's to any additional compensation, whether or not such change, directive or other communication from Client affects the means, methods, techniques or design of the Work or the Project.

15.2 In addition to Contractor's right to terminate or suspend the Work under §15.1, Contractor may, at any time terminate or suspend performance of the Work, or any portion thereof, at Contractor's convenience by service of written notice upon Subcontractor. Upon receipt of such notice, Subcontractor shall, unless the notice directs otherwise, immediately discontinue the Work, or designated portion thereof, and placing of orders for services, materials, facilities and supplies in connection with the performance thereof, and shall, unless otherwise directed, make every effort to procure cancellation of all existing orders or contracts related to such suspended or terminated Work upon terms approved by Contractor, or at the option of Contractor, give Contractor the right to assume the responsibilities and benefits of those obligations. Subcontractor shall thereafter do only such Work as may be necessary to preserve and protect the Work already in progress and to inventory and protect material and equipment on the job site, equipment in storage off of the job site (if previously allowed in writing on terms directed by Contractor) or in transit thereto.





Reference No.: 733700 Subcontract No.: 08.0040 Page 10 of 31

- 15.2.1 Upon such termination for Contractor's convenience, Subcontractor shall be entitled to payment in accordance with and subject to the requirements of Article 5.0, only as follows: (1) payment shall be made to Subcontractor commensurate with the percentage of Work properly completed through the date of termination (but in any event no more than one hundred fifteen percent (115%) of the direct cost of the Work completed); plus (2) such other actual and unavoidable out of pocket termination costs incurred by Subcontractor and approved in advance by Contractor.
- 15.3 Termination costs to be paid to Subcontractor under this Article 15.0 shall be subject to Contractor's right to withhold as set forth in Article 5.0.
- 16.0 DEFAULT BY SUBCONTRACTOR; TWO DAYS TO CURE: 16.1 Immediately following the request of Contractor, Subcontractor shall provide Contractor with assurances of its ability to perform the Work and to meet its obligations in accordance with the requirements of this Subcontract or, in lieu thereof, statements of credit worthiness, either of which shall be subject to the reasonable satisfaction of Contractor.
- 16.2 Contractor may terminate this Subcontract for default if Subcontract defaults under any obligation under this Subcontract and fails to cure said default within two (2) business days of receipt of notice of said default from Contractor. A default of Subcontractor's obligation(s) under this Subcontract shall include, without limitation, any of the following: (i) Subcontractor's failure to commence promptly or to prosecute the Work diligently in accordance with the Work Progress Schedule; (ii) Subcontractor's failure to provide Contractor with reasonably satisfactory evidence of its ability to perform (including evidence of its financial capability to perform) and/or statements of credit worthiness as required under §16.1; (iii) Subcontractor's insolvency, the filing by Subcontractor of a petition in bankruptcy, Subcontractor's making of an assignment for the benefit of creditors, or the filing of an involuntary petition for Subcontractor's bankruptcy by creditors; (iv) becoming involved in labor difficulties that do not constitute an excusable condition under §8.5; (v) Subcontractor's failure to maintain any policy of insurance or provide proof of such insurance as required under Article 10.0; (vi) Subcontractor's failure to pay taxes or payroll when due; (vii) Subcontractor's failure to timely pay or discharge in full or provide adequate security for the payment of all claims of any persons, or entities furnishing or claiming to have furnished labor, materials, equipment or incidentals used in, upon or for the Project, including any of Subcontractor's lower tier subcontractors or their suppliers, whether or not, as to any such claim, a lien or right of enforcement is established or attempted to be established upon or against the Project, the Work, the real property upon which the Work is situated, upon any bond furnished by Contractor or Client or upon any monies payable to Contractor by Client; and/or (viii) Subcontractor's material failure to perform any other obligation required of it under this Subcontr
- 16.3 In the event Subcontractor fails to cure any default under §16.2 within the time period stated in §16.2, Contractor, subject to Article 28, and in addition to its rights available to it under §16.2, shall have the right (but not the duty), if it so elects, to exercise any or all of the following remedies:
- 16.3.1 Supply such number of workers and quantity of materials, equipment and other facilities as Contractor deems reasonably necessary for the completion of the Work or the orderly progress of the Work, or any part of the Work that Subcontractor has failed to complete or perform, and charge the cost thereof to Subcontractor, who shall be liable for the payment of same including reasonable overhead, profit and actual attorneys' fees incurred as a result of Subcontractor's default;
- **16.3.2** Contract with one (1) or more additional contractors to perform such part of the Work as Contractor shall determine will provide the most expeditious completion of the total work required by the Contract and charge the cost thereof to Subcontractor;
- 16.3.3 Withhold payment of any monies due Subcontractor pending corrective action to that required by and to the reasonable satisfaction of Contractor; and/or
- 16.3.4 Terminate the right of Subcontractor to perform the Work and to use any materials, equipment, appliances or tools at the site of the Work furnished by or belonging to Subcontractor.
- 16.4 In the event of an emergency affecting or threatening to affect the safety of persons or property, Contractor may proceed as provided in §16.3.1 through §16.3.4 without notice required under §16.2.
- 16.5 In the event Subcontractor files a petition in Bankruptcy under Chapter 7, 11 or 13, the Parties agree that any delay attendant to the assumption or rejection of this Subcontract by a trustee or a debtor-in-possession will be prejudicial to Contractor. Consequently, Subcontractor hereby stipulates, in order to minimize delay to the Project and to mitigate damages and/or other prejudice suffered by Contractor, to a notice period of ten (10) calendar days for Contractor's motion to require Subcontractor to elect to assume or reject this Subcontract.





Reference No.: **733700** Subcontract No.: **08.0040** Page 11 of 31

16.6 Subcontractor hereby assigns, transfers, pledges and conveys to the Contractor (effective as of the date of this Subcontract, but only in the event of default, breach or failure by Subcontractor and subject to, and to the extent of, Contractor's acceptance of such assignment(s)), as collateral security, to secure the obligations under this Subcontract and any other indebtedness and liabilities of the Subcontractor to Contractor, all of Subcontractor's rights under this Subcontract, including Subcontractor's right, title and interest in and to (1) all subcontracts or supply contracts let by Subcontractor in connection therewith and such subcontractor's or supplier's surety bonds, (2) all machinery, plant, equipment, tools and materials which shall be on the site or sites of the work or elsewhere for the purposes of the Subcontract, including all materials ordered for this Subcontract, (3) any and all sums due or to become due on this Subcontract; and (4) any actions, causes of action, claims or demands whatsoever which the Subcontractor may have in any way arising out of or relating to this Subcontract.

17.0 RESPONSIBILITY FOR WORK: Subcontractor shall be responsible for and shall bear any loss of or damage to the Work and all materials, supplies and equipment under the care, custody and/or control or for which it is otherwise responsible until final approval and acceptance thereof by Client, except where such loss or damage results from the negligence of Contractor. Subcontractor shall be responsible to Contractor for any damage to the Work, other Project work or property of Client, Contractor or others caused by any act or omission of Subcontractor or anyone acting for or on its behalf.

18.0 LABOR CONDITIONS: Subcontractor shall not use any class of workers, materials, or methods which may cause strikes or labor disturbances and Subcontractor shall comply with all labor Laws applicable to the Work as are now or may hereafter be in force and effect. Subject to relief as allowed under the Contract (as provided in §8.5), Subcontractor shall be required to man its Work so as to maintain the Work Progress Schedule regardless of any labor activity including strikes, picketing or other labor disturbances. Subcontractor shall conform to prevailing wage and certified payroll requirements applicable to the Project, if any. Immediately upon the request of Contractor, Subcontractor shall remove from the Project any employee whom Contractor determines to be incompetent, undesirable or presents a threat of risk or harm to others. Subcontractor agrees that if any portion of the Work is further subcontracted, each such lower tier subcontractor shall be bound by and observe the provisions of this Article 18.0 to the same extent as herein required of Subcontractor, and that a copy of this Article 18.0 imposing such obligations upon such sub-subcontractor shall be included in any further sub-subcontract.

19.0 INSPECTION; ACCESS; TWO DAYS TO CORRECT; POSSESSION PRIOR TO COMPLETION: 19.1 The Work shall at all times be subject to inspection by Client and Contractor or their designated agents or representatives. Subcontractor shall at all times furnish to Contractor, its Client and their representatives safe and ample facilities for inspecting the Work (including materials) at the site of construction, shops, factories or any place of business of Subcontractor where the Work may be in the course of preparation, process, manufacture or treatment. Subcontractor, as often as required by Contractor, shall furnish to Contractor full reports of the progress of the Work at any place where the Work may be in the course of preparation or manufacture. Such reports shall show the progress of such preparation and manufacture in such details as may be required by Contractor, including any plans, drawings or diagrams in the course of preparation.

19.2 In the event that any part of the Work is determined by Client or Contractor to be improper, defective or otherwise fails to conform to the Subcontract Documents, either during the performance of the Work or during any guarantee period provided in the Contract or if there be none, then for a period of one (1) year from completion and acceptance of the Project by Client, Subcontractor shall, within two (2) business days of notification by Contractor to do so, immediately and diligently proceed to remove, dispose of and replace the same in a diligent manner and make good all work damaged or destroyed by or as a result of such defective, improper or nonconforming Work or materials or by the dismantling, removal or replacement of the defective, improper or non-conforming Work at Subcontractor's own cost and expense. If Subcontractor shall fail within such two (2) business day period to immediately proceed and continue thereafter to diligently replace or correct improper, defective or non-conforming Work promptly and completely, Contractor, at its option, may replace or correct the same and Contractor may declare Subcontractor in default of this Subcontract and shall have the right, if it so elects, to exercise any or all of the remedies in §16.3.1 through §16.3.4. Subcontractor shall be responsible for all costs, expenses (including actual attorney's fees), liabilities and consequential damages incurred by Contractor in connection with said replacements or corrections, whether removed, disposed of and replaced by Subcontractor or Contractor or others. Subcontractor shall execute and timely provide any special guarantees in the manner and to the extent required by the terms of the Subcontract Documents. Provided, however, the period of time given Subcontractor to repair and/or replace Work that is improper, defective or is otherwise non-conforming shall not be construed as granting to Subcontractor a time extension for the performance of all or any portion of the Work. Delays caused by Subcontractor's (i) failure to timely perform the Work in a proper manner or (ii) correction or replacement of improper, defective or otherwise non-conforming Work shall be the sole responsibility of Subcontractor.

19.3 Whenever it may be useful or necessary for Contractor or its Client to do so, Contractor or its Client shall be permitted to occupy and/or use any portion of the Work which has been either partially or fully completed by Subcontractor before final inspection and acceptance thereof by the Client, but such use and/or occupation shall not relieve Subcontractor of his guarantee of said Work nor of his obligation to make good at his own expense any defect in materials and/or workmanship which may occur or develop prior to Contractor's release from responsibility to the Client. The Subcontractor will not be held responsible for any damages found to be caused by the sole negligence of Contractor or its Client.





Reference No.: **733700** Subcontract No.: **08.0040** Page 12 of 31

20.0 CLEAN-UP: Subcontractor shall carry on its operations so that the Project site shall at all times be clean, orderly and free from debris and waste. Upon completion of its Work, Subcontractor shall remove all equipment and unused materials from the site, clean up all refuse and debris, and leave the site of the Work clean, orderly and in good condition. If Subcontractor fails to keep the premises clean in accordance with the requirements of this Article 20.0, Contractor may clean, or cause the premises to be cleaned by others and shall be reimbursed by Subcontractor for all costs and expenses incurred in doing so plus a reasonable amount for overhead and profit.

- **21.0 BOND**: Contractor may, at Contractor's sole discretion and expense, require that Subcontractor obtain and furnish surety bonds guaranteeing full performance of this Subcontract and that Subcontractor will promptly and fully pay for all work, labor and materials and other charges or costs in connection with the Work (collectively the "Bonds"). The form of the Bonds, the amount of the Bonds and surety underwriting the Bonds shall be acceptable to Contractor in its sole discretion.
- **22.0 ENTIRE SUBCONTRACT: 22.1** The Subcontract Documents comprise the entire and integrated agreement between Contractor and Subcontractor and supersede any and all prior negotiations, representations, understandings, and agreements, written or oral, relating to the Work or the relationship between the parties. No agent of either party has authority to make and the parties shall not be bound by or liable for any statement, representation, promise or agreement not set forth herein. No changes, amendments or modifications of the terms hereof shall be valid unless reduced to writing and signed by Contractor and Subcontractor.
- 22.2 If, following execution of this Subcontract, the Parties enter into any subsequent agreement or understanding pertaining to or relating to the Project or Work to be performed on the Project, such agreement or understanding shall be governed by the terms and conditions of this Subcontract notwithstanding the existence of any pre-printed terms and conditions or oral agreement and the terms and conditions of this Subcontract shall govern in all cases.
- 23.0 DISPUTE RESOLUTION: 23.1 Any controversy, claim or dispute of whatever nature and on any theory, including negligence, misrepresentation, fraud or construction defect, that arises out of or relates to the Subcontract (including its interpretation or application), the Work, Sub Claims or any acts or omissions of Subcontractor, including the determination of the scope of this agreement to arbitrate, shall be decided by binding arbitration in accordance with the then current Construction Industry Rules of the American Arbitration Association, except for: disputes that have been waived by the making or acceptance of final payment or questions regarding the licensure of Subcontractor or litigation referred to in §23.2.
- 23.1.1 Prior to demanding arbitration, the Subcontractor shall notify Contractor of its intent to demand arbitration and promptly upon such notification; the Contractor and Subcontractor shall meet to informally resolve such dispute. In the event that no resolution is achieved, and if, but only if, the parties mutually agree, then prior to the Subcontractor demanding arbitration, the parties shall make a good faith effort to resolve the dispute by negotiations between representatives with decision-making power, who, to the extent possible, shall not have had substantive involvement in the matters of the dispute.
- 23.1.2 Notice of the demand for arbitration shall be filed in writing with the other party to this Subcontract in accordance with the then current Construction Industry Rules of the American Arbitration Association. For claims arising out of or related to Sub Claims or Claims For Extras (as defined in §23.4) the demand for arbitration may be made by either party and shall be made within a reasonable time but in no event later than thirty (30) calendar days after final acceptance of the Project by Client. All other demands for arbitration, including warranty claims and joinder or consolidation into other arbitration proceedings under §23.1.3 and §23.3, may be made by either party and shall be made within a reasonable time but in no event after the date when institution of legal or equitable proceedings based on such claim, dispute or other matter in question would be barred by the applicable statute of limitations. The award rendered by the arbitrator(s) shall be final and judgment may be entered upon it in accordance with applicable Law in any court having jurisdiction.
- 23.1.3 Contractor may consolidate any arbitration between Contractor and Subcontractor or join Subcontractor into any arbitration or other dispute resolution proceeding provided for in the Contract ("Consolidated Arbitration") to the extent permitted by the Contract. Provided Subcontractor is afforded a reasonable opportunity to participate in the Consolidated Arbitration, Subcontractor shall be bound by the result of the Consolidated Arbitration to the same degree as the Contractor including the award rendered by the arbitrator(s) being final and judgment allowed to be entered upon it in accordance with applicable Law in any court having jurisdiction.
- 23.2 In the event any person or entity including Client is engaged in litigation ("Litigation") against Contractor, or against Client who is pursuing a cross action against Contractor, on any theory of claim including negligence, fraud, strict liability or construction defects arguably caused in whole or in part by Subcontractor, then Contractor may join Subcontractor into such litigation.
- 23.3 If Contractor agrees or is compelled to arbitrate said Litigation or participate or engage in another form of dispute resolution ("Third Party Arbitration"), then Contractor may consolidate any arbitration or other proceeding between Contractor and Subcontractor or join Subcontractor into any such Third Party Arbitration. Provided Subcontractor is afforded a reasonable opportunity to participate in the Third Party Arbitration, Subcontractor shall be bound by the result of the Third Party Arbitration, including any and all decisions, findings or





Reference No.: **733700** Subcontract No.: **08.0040** Page 13 of 31

determinations made thereunder to the same degree as the Contractor is bound by such result including the arbitrator(s) award being final and judgment allowed to be entered upon it in accordance with applicable Law in any court having jurisdiction.

- 23.4 The dispute resolution procedure, if any, set forth in the Contract is specifically incorporated herein and made a part of this Subcontract to the extent it is applicable to claims for time or compensation that may originate with or may otherwise be the responsibility of Client. The Subcontractor agrees to make any Claims For Extras in the same manner as required in the Contract for like claims of the Contractor upon the Owner, and in such times as will enable the Contractor to timely present such claims to the Owner for payment or recognition. To the extent that a Sub Claim arguably originates with Client or other third parties or for which Client may be responsible, Contractor in its sole discretion, shall have the right to (i) prosecute such Sub Claim on behalf of Subcontractor or (ii) require Subcontractor to prepare, present and prosecute the Sub Claim to the extent it relates to such Sub Claim (collectively "Claims For Extras"). If Contractor elects to prosecute a Claim for Extras under §23.4(i), Subcontractor shall cooperate fully with the Contractor and to furnish all documents, statements, witnesses and other information required by Contractor for such purpose and to pay or reimburse Contractor for all expenses and costs, including reasonable attorneys' fees, incurred in connection therewith to the extent of the Subcontractor's interest in a claim or dispute. Where Contractor elects to have Subcontractor prosecute the Claim For Extras under §23.4(ii), Contractor's sole obligation shall be to present any approved and timely-filed Claims For Extras to Client in accordance with the procedure provided in the Contract. Contractor will not be liable to Subcontractor on account of any Sub Claims not timely or properly presented by Subcontractor and, subject to the other provisions of this Subcontract, shall be obligated to pay to Subcontractor only the proportionate share of any sums paid by the Client for the Work of Subcontractor or if an amount is not specifically delineated to be paid to Subcontractor, than an amount to which Subcontractor is reasonably entitled. Contractor's obligation under §23.4(i) or §23.4(ii) shall be subject to Subcontractor's compliance with all applicable Laws including those relating to false claims, dispute and claim certifications, cost and pricing data requirements and such other requirements as deemed necessary by Contractor to determine the validity of the Claim For Extras. To the extent required by Contractor, Subcontractor shall promptly provide to Contractor (i) all written documentation of cost and pricing data and any other information Subcontractor intends to rely upon in support of the Sub Claim; and (ii) written certification or other verification attesting to the validity and accuracy of the Sub Claim in form and content required by Contractor, Law and/or Client. The timely delivery to Contractor of the documentation and certification required is an express condition precedent to Contractor's obligation to handle, present, process and/or prosecute the Sub Claim.
- 23.5 It is further specifically agreed by the parties hereto that no claim, dispute or controversy shall interfere with the progress and performance of the Work and, unless otherwise agreed in writing, Subcontractor shall carry on the Work and maintain the Work Progress Schedule as directed by Contractor pending resolution of any dispute, including arbitration, and, if so, Contractor shall continue to make payment of undisputed amounts due to Subcontractor in accordance with this Subcontract.
- 23.6 This Article shall not be deemed a limitation of any rights or remedies which Contractor and/or Subcontractor may have under any federal or Guam mechanics' lien Laws or under any applicable labor and material payment bonds unless such rights or remedies are expressly waived by it. If required by statute to perfect or preserve lien, bond or stop notice rights, Subcontractor may commence litigation. However, such litigation shall be stayed or enjoined by stipulation or upon Contractor's motion pending a resolution or adjudication of any underlying issues in dispute that affect payment to Subcontractor.
- 23.7 This Article shall survive any actual or purported termination or rescission of this Subcontract.
- 23.8 If Contractor or Subcontractor brings a motion to dismiss or stay litigation pending arbitration or action or motion to compel arbitration, the prevailing party in such proceeding shall be entitled to recover reimbursement of actual attorney and expert fees and costs incurred to prosecute the motion.
- 24.0 MISCELLANEOUS: (i) As used in this Subcontract, the singular includes the plural; the masculine includes the feminine and neuter; (ii) If more than one person or entity is named herein as Subcontractor, their obligations shall be joint and several; (iii) All references to Subcontractor shall be construed to include its agents, employees, and its subcontractors of every tier and their vendors unless the context indicates otherwise; (iv) All references to sub-subcontractor shall include its subcontractors of every tier and their vendors unless the context indicates otherwise; (v) The parties and their respective assigns (to the extent such assigns are consented to as required in §13.2), legal representatives and other successors in interest will be bound hereby; (vi) The term "including" shall be construed to mean "including, but not limited to"; For the purposes of this Subcontract, the Territory of Guam is deemed a "state."
- **25.0 PRECAUTIONS: 25.1** Subcontractor shall be solely responsible for providing a safe place to work for Subcontractor's employees and for employees of its subcontractors and suppliers (hereafter "Employees"). Subcontractor shall take all necessary steps to protect Employees and others exposed to the operations of Subcontractor from risk of harm or injury including any conditions that present a risk of occupational injury or illness to Employees. Subcontractor shall not be relieved of such responsibility by any acts or omissions of Contractor. Subcontractor shall continuously inspect its Work, methodologies and materials (including the adequacy of and required use of all safety equipment) to discover and determine any conditions which pose a risk of bodily harm or injury to Employees or other persons on site or a risk of damage to property (including, the property of Client, Contractor and/or third parties) or a violation of any applicable Laws. With





Reference No.: **733700** Subcontract No.: **08.0040** Page 14 of 31

regard to industrial safety, Subcontractor's obligations are independent obligations and Subcontractor is vested with the authority and shall take all precautions which are necessary and adequate to protect against and/or correct any such conditions immediately upon first becoming aware of the existence of such conditions whether by Subcontractor's discovery of such conditions or being made aware of the existence of such conditions by others.

- 25.2 Subcontractor shall receive, respond to, defend and be responsible for and shall indemnify, defend and hold Client and Contractor harmless from, and against, any citation, order, claim, charge or action arising out of, resulting from or incidental to conditions that present a risk of occupational injury or illness to Employees regardless of whether such condition results from the acts or omissions of Subcontractor (or those for whom it is responsible) or are a contributing cause to the conditions giving rise to or resulting in the issuance or prosecution of a citation, order, claim, charge or action.
- 25.3 Subcontractor's obligations under this Article 25.0 are in addition to and not in lieu of all other provisions of this Subcontract including its obligations to comply with Laws as provided in Article 9.0. Subcontractor shall adopt a written accident prevention program (hereafter "APP") that establishes minimum safety standards for Subcontractor's managers, supervisory personnel and employees. Where an APP is prescribed by Law, the APP shall comply with such Law. The APP shall (i) assign responsibilities for its managers, supervisory personnel and employees, (ii) establish standard procedures for hazard evaluation and employee and supervisor training for hazard recognition, (iii) provide for consistent, disciplined APP program enforcement, (iv) establish accident investigation procedures, (v) include record keeping to document Subcontractor's compliance with the requirements of the APP, and (vi) identify and list the minimum accident prevention requirements for typical operations. The foregoing shall be required as part of the APP regardless of whether required by Law. Nothing contained herein shall relieve Subcontractor of responsibility for safety and/or a safe place to work to the extent the same is required by the Contract. If required by Contractor, Subcontractor shall promptly provide Contractor with a copy of its APP.
- **26.0 SEVERABILITY**: It is the Parties intent that this Subcontract not violate any Law. In the event that any provision of this Subcontract shall at any time contravene in whole or in part any applicable Law, then such provision shall remain in effect only to the extent permitted, and the remaining provisions hereof shall remain in full force and effect.
- **27.0 EQUAL OPPORTUNITY**: Subcontractor will comply with all provisions of Executive Order No. 11246, of September 24, 1965, or as amended or superseded, and of the rules, regulations and relevant orders of the Secretary of Labor and other rules and regulations as provided by any Law. Subcontractor shall furnish all information and reports as may be required and will permit access to Subcontractor's books, records, and accounts by Contractor, governmental agencies and Secretary of Labor for the purpose of investigating compliance with such rules, regulations and relevant orders.
- **28.0 LEGAL REMEDIES**: Either party hereto shall be entitled to all remedies afforded by Law or in equity to enforce their respective rights under this Subcontract. With the exception of the obligation of the Parties to proceed with arbitration as provided in Article 23, no right or remedy in the Subcontract, including those remedies provided for in Articles 16 and 19, are intended to be exclusive of any other right or remedy, but every such right or remedy shall be cumulative and shall be in addition to and not a limitation of any duties, obligations, rights and remedies otherwise imposed or available by Law, in equity or in the Subcontract.
- **29.0 JURISDICTION**: It is understood and agreed that each and every provision of this Subcontract, including any alleged breach thereof, shall be interpreted in accordance with the Laws of Guam and the Laws of the United States of America applicable to Guam.
- **30.0 TAX LEVIES:** Contractor may, in its absolute discretion, pay directly to taxing authorities any sums otherwise due Subcontractor and not subject to set-off under §5.7 hereof, upon receipt of a tax levy from such taxing authority.
- **31.0 COMMUNICATIONS:** Subcontractor shall not communicate directly with Client or design professionals not under contract to Subcontractor. All communications to be addressed to Client or design professionals not under contract to Subcontractor shall be through Contractor only.
- **32.0 NO WAIVER:** No waiver of any rights hereunder shall be valid or binding upon the parties unless the same be in writing and signed by authorized representatives of the parties. Failure of Contractor to insist upon strict performance of the Subcontract, or to exercise any option herein, shall not be construed to be a waiver of Contractor's right to insist on strict performance of any part of the Subcontract.
- **33.0. WRITTEN NOTICE: 33.1** Whenever, under the terms of this Subcontract, it becomes necessary, appropriate, or desirable for a party to give notice to the other party, said notice shall be in writing and shall be personally delivered or given or mailed by commercial overnight courier service or by registered or certified United States mail return receipt requested, with first-class postage prepaid, addressed as follows:







Reference No.: **733700** Subcontract No.: **08.0040** Page 15 of 31

To Contractor: GRANITE-OBAYASHI a Joint Venture

701 East Main Street Lewisville, TX 75057

Attn: Glenn Walsh, Project Executive

To Subcontractor: Relyant Global, LLC 335

High Street

Maryville, TN 37804 P: 865-984-1330 Attn: Alix King

Relyant Global, LLC

125 Tun Jesus Crisostomo Street

Suite 206

Tamuning, GU 96913 P: 671-989-0941 Attn: Kim Duenas

33.2 At any time, either party may designate a new or different address to which notices are to be sent, which notice of a new or different address shall be given as hereinabove immediately provided.

33.3 Any notice shall be deemed given as of the time that the same is personally delivered to the intended recipient or as of a date two (2) days (excluding weekends and federal and state holidays) later than the time that the same is properly deposited in the United States mail, if such notice deposited in the United States mail is given as hereinabove provided, or as of a date one day (excluding weekends and federal and state holidays) later than the time that the same is properly given to a nationally recognized commercial overnight courier service, for delivery to the other Party.

IN WITNESS WHEREOF, the Parties hereto have executed this Subcontract the day and year first above written.

RELYANT GLOBAL, LLC SUBCONTRACTOR	GRANITE-OBAYASHI a Joint Venture CONTRACTOR
By: AA91FC839F7C4AE Title: Director of Contracts Name (Type) Chris Twiner	By Matt typer DBE86E10C06F467 Title: Senior Vice President Federal Group Name: Mathew Tyler
☐ Corporation ► Partnership	
☐ Proprietorship ☐ Other	
Subcontractor's State License No.	
Date of Expiration	
Federal I.D. No.	





Reference No.: **733700** Subcontract No.: **08.0040** Page 16 of 31

ATTACHMENT A.1

WORK TO BE PERFORMED:

Item	Cost Code	Description	Quantity	U.M.	Unit Price	Total
No.						
001		Project Planning and Administration	1	LS	\$16,723.80	\$ 16,723.80
002		MEC Clearance Work Plan	1	LS	\$8,205.50	\$ 8,205.50
003		Mobilization/Demobilization	1	LS	\$95,754.90	\$ 95,754.90
005		Anomaly Investigation, Surface Clearance &				\$ 0.00
		EZ Control/ Signage				
005.1		OPTION - AREA A	15	AC	\$532,491.45	\$ 0.00
005.2		OPTION - AREA B	35	AC	\$1,242,480.05	\$ 0.00
005.3		OPTION - AREA D	35	AC	\$1,1242,480.05	\$ 0.00
005.4		AREA F	28	AC	\$993,984.04	\$ 993,984.04
006		After Action Report	2	EA	\$5,172.08	\$ 10,344.00
007		AA Support/Spotter Daily Rate	5,540	HR	\$106.91	\$ 592,281.40
800		Bond	1.1	%	\$1,717,293.64	\$ 18,890.23
			TOTAL			\$ 1,736,183.87

AREAs A, B, and D will be additional options once modification is processed, this will also add additional After Action Reports and Spotter Hours. The quantities above are estimates and unit prices will govern.





Reference No.: **733700**Subcontract No.: **08.0040**Page 17 of 31

ATTACHMENT A.2

SCOPE OF WORK

- Subcontractor is responsible for MEC clearance as well as the associated preconstruction submittals in accordance with specification 01 57 19.03 SUPPLEMENTAL TEMPORARY ENVIRONMENTAL CONTROLS, JOINT REGION MARIANAS EXPLOSIVES SAFETY SUBMISSION (JRM ESS) and including the items below
 - A. MEC production shall match or exceed earthwork production for an average or 37,600 square meter per shift for Spread and Scan Clearance to a Target of Interest (TOI) of 20mm and 12,500 square meters per shift for E1-16B for a TOI of 60mm and Full Clearance to a TOI of 20mm.
 - B. MEC Clearance will minimize impacts on earthwork operations by working night shifts if necessary, while supporting the operations during the day with UXO Spotters.
 - C. UXO Spotters will observe E1-16B soil moving operations using armored equipment, materials, or other methods in compliance with the JRM ESS.
 - D. Preparation of ESS Work Plans specified in 01 57 19.03 paragraph 1.4.1 ANOMALY AVOIDANCE WORK PLAN (WP), paragraph 1.4.2 DIGITAL GEOPHYSICAL MAPPING (DGM) WORK PLAN (WP), and paragraph 1.4.3 ANOMALY INVESTIGATION WORK PLAN (WP)
 - E. Traffic Control and notifications shall be in accordance with specification 01 50 00 TEMPORARY CONSTRUCTION FACILITIES AND CONTROLS and 01 57 19.03 Paragraph 1.5 Installation Commanding Officer (ICO) Notification.
 - F. Provide up to six (6) After Action Reports as required by NOSSA and specification 01 57 19.03 paragraphs 1.9.1 Anomaly Avoidance After Action Report (AAR), paragraph 1.9.2 Digital Geophysical Mapping (DGM) After Action Report (AAR), and paragraph 1.9.3 Anomaly Investigation After Action Report (AAR).
 - G. Providing traffic control barricades/signage, road guards and flagmen in accordance with MUTCD 6E and specification 01 50 00 TEMPORARY CONSTRUCTION FACILITIES AND CONTROLS.
 - H. Participation in pre-project support and coordination meetings between GOJV and NAVFAC.
 - I. Daily Reports, QA/QC Reports, and inspection reports shall be provided the following business day by 1000 hrs (ChST).
 - J. Providing qualified personnel for the duties of UXOSO, UXOQCS, SUXOS, Geophysicist (if needed) and UXO Technicians as detailed in specification 01 57 19.03 paragraph 1.3 PERSONNEL QUALIFICATIONS AND DUTIES. Personnel shall be approved by NAVFAC as required.
 - K. Equipment needed to support operations.
 - L. Community Outreach if required as described in specification 01 57 19.03 paragraph 3.3 Community Outreach.
 - M. Exclusion Zone Enforcement as described in specification 01 57 19.03 paragraph 3.4.
 - N. Follow all procedures and reporting of MEC/MPPEH in accordance with specification 01 57 19.03 paragraphs 1.7and 3.7.
 - O. Provide UXOQCS personnel to accompany Government Quality Assurance inspections.

2. Scope Exclusions:

- A. Subcontractor is not responsible for:
 - i. Equipment for Mass Excavation and Disposal of Excavation Soil will be provided by GOJV
 - ii. Light plant staging and fuel for night operations
 - iii. Archaeological Monitoring per Specification Section 01 57 19.01 20 paragraph 3.5.
 - Preconstruction survey and Layout of Areas per Specification Section 01 57 19.00 20 paragraph 1.6.1
 - Fencing per Specification Section 01 50 00.05 20 paragraph 3.1.1.2 and Temporary Environmental Controls for Open Excavation
 - vi. GOJV will coordinate jobsite access in accordance with 01 14 00.05 20 WORK RESTRICTIONS FOR DESIGN-BUILD paragraph 1.4.1.2 IDENTIFICATION BADGES and paragraph 1.5 SECURITY REQUIREMENTS. It is the subcontractor's responsibility to provide personnel that qualify for base access. No contractor caused delays will occur once subcontractor has mobilized.
 - vii. GOJV will coordinate and pay for all utility outages required to complete intrusive MEC activities
 - viii. GOJV will provide all permits and utility locate/clearances prior to mobilization.
 - ix. Damage to buried utilities that are not marked by locator in accordance with 01 35 26.05 20 Government Safety Requirements for Design-Build paragraph 3.6 Excavations.
 - x. MEC/MPPEH encountered will be transported or disposed of by EOD per section 01 57 19.03 paragraph 3.9 REQUIREMENTS WHEN MEC/MPPEH IS ENCOUNTERED.
 - xi. Site grading or final restoration after the intrusive investigation is complete.
 - xii. Soil erosion or temporary environmental controls after the work is performed.





Reference No.: **733700** Subcontract No.: **08.0040** Page 18 of 31

- 3. Contractor will provide the following at no cost for Work under this Subcontract:
 - A. Contractor will provide Subcontractor with one (1) week notice prior to the start of Subcontractor's initial Work operations and other phased Work. Once a continuous or phased operation has begun, Subcontractor is expected to stay fully mobilized on the Project site, with qualified management and supervision including persons with authority to make all decisions necessary for the day-to-day operations of the Work.
- 4. All Work and payment under this Subcontract shall be in accordance to the Contract Drawings and Specifications, SOW, Special Provisions, and Amendments.
- Subcontractor shall submit regular schedule updates, including, but not limited to, daily and weekly work plans and weekly Three-Week Look-a-head schedules.
- 6. SUBCONTRACT EXECUTION CONTINGENT This Subcontract is not legally effective unless and until Subcontractor is approved by Client to perform the Work described, if such approval is required under the Contract.
- 7. DAILY REPORT Subcontractor shall furnish a Daily Report to Contractor the first workday following the day the Work is performed. The Daily Report shall include, but not be limited to, the following items:
 - A. Number of employees on the job
 - B. Description and number of pieces of equipment on job
 - C. Description and location of Work performed
 - D. Quantities of Work performed
 - E. Verbal instructions received from Contractor
 - F. Requests made by Subcontractor
 - G. Specific problems that need resolution
 - H. Situations now delaying or slowing progress
 - I. Miscellaneous remarks
- 8. BOND Subcontractor shall furnish a satisfactory payment and performance bond on the form provided by Contractor in the full amount of Subcontract. Subcontractor shall cause the bond premium to be billed directly to Contractor by bonding agent. Contractor will deduct from payments due Subcontractor any bond premium invoiced in excess of 1.0%. In no event will Contractor pay more than actual invoice bond premium.
- 9. CERTIFIED PAYROLLS Original and one copy of Weekly Certified Payrolls must be submitted to the Contractor within seven (7) days of the ending date of the payroll period. Project descriptions and contract numbers must be listed on all documents. Final payroll report must be marked "FINAL". Failure to submit accurate Certified Payrolls as specified may result in withholding of payment from Subcontractor. Provided, however, and notwithstanding the preceding sentence, in no event shall the amount withheld from Subcontractor be less than the amount, if any, withheld from Contractor by Owner as a result of the failure of Subcontractor to timely provide accurate and correct Certified Payrolls.
- 10. CERTIFIED PROGRESS PAYMENTS Subcontractor shall fully comply with FAR clause 52.232-5 Payments Under Fixed-Price Construction Contracts for all payments. Progress payments shall include the Contractor Certification clause required by FAR clause 52.232-5 (below).

I hereby certify, to the best of my knowledge and belief, that --

- (1) The amounts requested are only for performance in accordance with the specifications, terms, and conditions of the contract;
- (2) All payments due to subcontractors and suppliers from previous payments received under the contract have been made, and timely payments will be made from the proceeds of the payment covered by this certification, in accordance with subcontract agreements and the requirements of chapter 39 of Title 31, United States Code;
- (3) This request for progress payments does not include any amounts which the prime contractor intends to withhold or retain from a subcontractor or supplier in accordance with the terms and conditions of the subcontract; and
- (4) This certification is not to be construed as final acceptance of a subcontractor's performance.





Reference No.: **733700**Subcontract No.: **08.0040**Page 19 of 31

(Name)		
(Title)		
(Date)		

- 11. PROMPT PAYMENT FOR SUBCONTRACTORS: All subcontractor payments shall be in accordance to FAR clause 52.232-27 PROMPT PAYMENT FOR CONSTRUCTION CONTRACTS.
 - A. Contractor will pay the subcontractor for satisfactory performance under its subcontract no later than 7 days from receipt of payment out of such amounts as are paid to the Contractor under this contract.
 - B. Contractor shall pay to the subcontractor an interest penalty for each payment not made in accordance with the payment clause
 - i. For the period beginning on the day after the required payment date and ending on the date on which payment of the amount due is made; and
 - ii. Computed at the rate of interest established by the Secretary of the Treasury, and published in the Federal Register, for interest payments under 41 U.S.C. 7109 in effect at the time the Contractor accrues the obligation to pay an interest penalty.
 - iii. Subcontractor shall flow down the FAR Clause 52.232-27 to each lower-tier subcontractor including the payment and interest penalty clauses in subparagraphs (c)(1) and (c)(2); and
 - iv. Subcontractor shall require each of its subcontractors to include such clauses in their subcontracts with each lower-tier subcontractor or supplier.
- 12. CONSTRUCTION WAGE DETERMINATION All Subcontractors shall provide certified weekly payroll reports as mandated by the contract documents and FAR 52.222-8 Payrolls and Basic Records (MAY 2014) in compliance with the following: GENERAL DECISION NUMBER: GU170002, Modification No. 0, Publication Date 07/07/2017

STATE/TERRITORY: Guam CONSTRUCTION TYPE: Heavy COUNTIES: Guam Statewide MODIFICATION NUMBER: 0 PUBLICATION DATE: 07/07/2017

- 13. FEDERAL ACQUISITION REGULATIONS (FAR) This Subcontract Agreement is subject to all applicable FAR requirements specified in Attachment A.5 and referenced in the Construction Contract, Specifications, SOW, Special Provisions, and Amendments. Subcontractor has reviewed and is familiar with the FAR requirements that pertain to Federal projects. Subcontractor acknowledges that all references to "Contractor" through the FAR requirements shall be interpreted as meaning this "Subcontractor" and all references to the "Subcontractor" throughout the FAR requirements shall be interpreted as meaning lower tier subcontractors.
- 14. Signed Statement and Acknowledgement Standard Form (SF) 1413 must be submitted within three (3) days from issuance date of this contract. The SF 1413 must identify all 2nd-Tier subcontractors. If there are 2nd-Tier subcontractors, those subcontractors must additionally provide signed SF 1413s. If during the project, a subcontractor is added, a revised and signed SF1413 must be submitted.
- 15. Subcontractor fully comply with FAR 52.219-8 UTILIZATION OF SMALL BUSINESS CONCERNS (OCT 2014) and FAR 52.219-9 SMALL BUSINESS SUBCONTRACTING PLAN (OCT 2015)
 - A. Subcontractor will adopt a subcontracting plan that complies with FAR Clause 52.219-9 when subcontract value is in excess of \$700,000 (\$1.5 million for construction of any public facility).
 - B. Subcontractors with subcontracting plans must submit the ISR and/or the SSR using the eSRS as per Far Clause 52.219-9.
 - C. When submitting an ISR and/or SSR, subcontractor shall provide the prime contract number, its own DUNS number, and the email address of the subcontractor's official responsible for acknowledging receipt of or rejecting the ISRs





Reference No.: **733700** Subcontract No.: **08.0040** Page 20 of 31

ATTACHMENT A.3

INSURANCE LIABILITY LIMITS – APPLICABLE TO ALL SUBCONTRACTORS OF ANY TIER:

1. Workers' Compensation & Employers' Liability

Workers' Compensation Limits - Statutory as required by applicable State Law

Employers' Liability Limits

\$1,000,000 Each Accident

\$1,000,000 Disease - Policy Limit

\$1,000,000 Disease - Each Employee

If exposure exists, Subcontractor must show evidence of USL&H coverage and Maritime coverage with a minimum limit of \$2,000,000 for Jones Act Coverage;

The insurer shall agree to waive all rights of subrogation against Contractor, its officers and employees arising from work performed by Subcontractor for Contractor.

2. Commercial General Liability (Occurrence Form Only) Limits

\$1,000,000 Each Occurrence

\$1,000,000 Personal Injury Liability

\$1,000,000 Aggregate for Products-Completed Operations

\$2,000,000 General Aggregate

Including products, completed operations, contractual liability, broad form property damage, independent contractors, and coverage for explosion, collapse, and underground damages. The General Aggregate shall apply on a per-project basis.

3. Automobile Liability Limits

\$1,000,000 Combined Single Limit Each Occurrence Bodily Injury and Property Damage Including Owned, Non-owned, and Hired Vehicles.

4. Professional Liability Limits – <u>APPLICABLE TO ALL SUBCONTRACTORS OF ANY TIER PERFORMING ANY PROFESSIONAL SERVICES INCLUDING ENGINEERS, SURVEYORS, OR CONSULTANTS.</u>

Professional Liability Limits

Limits for Professional Services of a non-critical nature (including access platforms, mix-designs, haul roads, surveyors, water and air systems, etc.):

\$1,000,000 Each Claim \$1,000,000 Aggregate

Limits for Professional Services of a critical nature (including design or engineering of permanent structures or temporary structures used in support of construction (including excavation) such as falsework or shoring and crane lifting):

\$5,000,000 Each Claim \$5,000,000 Aggregate

5. Contractor's Pollution Liability Limits – APPLICABLE TO ALL SUBCONTRACTORS OF ANY TIER PERFORMING ANY REMOVAL, REMEDIATION, ABATEMENT, TRANSPORTATION OR DISPOSAL OF ANY HAZARDOUS MATERIALS.

\$1,000,000 Each Occurrence \$1,000,000 Aggregate

Claims Made polices shall include a three-year extended reporting period beyond completion of the Work.

6. EQUIPMENT INSURANCE is the responsibility of and must be maintained by all Subcontractors of any tier for Subcontractor owned, leased, rented and borrowed equipment.





Reference No.: **733700** Subcontract No.: **08.0040** Page 21 of 31

The additional insured endorsement required under § 10.3 shall be written utilizing standard ISO form CG 20 10 04 13 in combination with ISO form CG 20 37 04 13, or the equivalent, as determined by Contractor, and the endorsement required under § 10.3 for primary and non-contributory coverage shall be written utilizing standard ISO form CG 20 01 04 13. There shall be no modification or change from the standard ISO form language.

The additional insured endorsement required under § 10.3 shall list the following as additional insured partied GRANITE-OBAYASHI a Joint Venture NAVFAC ROICC Finegayan

Subcontractor shall fully comply with FAR 52.228-5 Insurance - Work on a Government Installation (Jan 1997), stated below.

- (a) The Contractor shall, at its own expense, provide and maintain during the entire performance of this contract, at least the kinds and minimum amounts of insurance required in the Schedule or elsewhere in the contract.
- (b) Before commencing work under this contract, the Contractor shall notify the Contracting Officer in writing that the required insurance has been obtained. The policies evidencing required insurance shall contain an endorsement to the effect that any cancellation or any material change adversely affecting the Government's interest shall not be effective –
- (1) For such period as the laws of the State in which this contract is to be performed prescribe; or
- (2) Until 30 days after the insurer or the Contractor gives written notice to the Contracting Officer, whichever period is longer.
- (c) The Contractor shall insert the substance of this clause, including this paragraph (c), in subcontracts under this contract that require work on a Government installation and shall require subcontractors to provide and maintain the insurance required in the Schedule or elsewhere in the contract. The Contractor shall maintain a copy of all subcontractors' proofs of required insurance and shall make copies available to the Contracting Officer upon request.





Reference No.: **733700** Subcontract No.: **08.0040** Page 22 of 31

ATTACHMENT A.4

PRELIMINARY INFORMATION:

Project: J-001B Finegayan Utilities and Site Improvements Phase I

GCCO Job No.: 733700

Date: September 14, 2020

Client's (Owner) Name: NAVFAC Pacific

Subcontractor: Relyant Global





Reference No.: 733700

Subcontract No.: 08.0040 Page 23 of 31

FAR CLAUSES INCORPORATED BY REFERENCE

DOCUMENT 00700 CONTRACT CLAUSES TABLE OF CONTENTS

- 52.252-2 Clauses Incorporated by Reference (FEB 1998)
- 52.202-1 Definitions (NOV 2013)
- 52.203-3 Gratuities (APR 1984)
- 52.203-5 Covenant Against Contingent Fees (MAY 2014)
- 52.203-6 Restrictions on Subcontractor Sales to the Government (SEP 2006)
- 52.203-7 Anti-Kickback Procedures (MAY 2014)
- 52.203-8 Cancellation, Rescission, and Recovery of Funds for Illegal or Improper Activity (MAY 2014)
- 52.203-10 Price or Fee Adjustment for Illegal or Improper Activity (MAY 2014)
- 52.203-12 Limitation on Payments to Influence Certain Federal Transactions (OCT 2010)
- 52.203-13 Contractor Code of Business Ethics and Conduct (APR 2010)
- 52.203-16 Preventing Personal Conflicts of Interest (DEC 2011)
- 52.203-17 Contractor Employee Whistleblower Rights and Requirement to Inform Employees of Whistleblower Rights (APR 2014)
- 52.204-2 Security Requirements (AUG 1996) Alternate II (APR 1984)
- 52.204-4 Printing or Copied Double-sided on Recycled Paper (MAY 2011)
- 52.204-9 Personal Identity Verification of Contractor Personnel (JAN 2011)
- 52.204-10 Reporting Executive Compensation and First-Tier Subcontract Awards (JUL 2013)
- 52.204-13 System for Award Management Maintenance (JUL 2013)
- 52.204-14 Service Contract Reporting Requirements (JAN 2014)
- 52.204-18 Commercial and Government Entity Code Maintenance (NOV 2014)
- 52.204-19 Incorporation by Reference of Representations and Certifications (DEC 2014)
- 52.209-6 Protecting the Government's Interest when Subcontracting with Contractors Debarred, Suspended, or Proposed for Debarment (AUG 2013)
- 52.209-9 Updates of Publicly Available Information Regarding Responsibility Matters (JUL 2013)
- 52.209-10 Prohibition on Contracting with Inverted Domestic Corporations (DEC 2014)
- 52.211-10 Commencement, Prosecution, and Completion of Work (APR 1984)
- 52.211-15 Defense Priority and Allocation Requirements (APR 2008)
- 52.211-18 Variation in Estimate Quantities (APR 1984)
- 52.215-2 Audit and Records—Negotiation (OCT 2010)
- 52.215-10 Price Reduction for Defective Cost or Pricing Data (AUG 2011)
- 52.215-11 Price Reduction for Defective Cost or Pricing Data Modifications (AUG 2011)
- 52.215-12 Subcontractor Cost or Pricing Data (OCT 2010)
- 52.215-13 Subcontractor Cost or Pricing Data Modifications (OCT 2010)
- 52.215-15 Pension Adjustments and Asset Reversions (OCT 2010)
- 52.215-18 Reversion or Adjustment of Plans for Postretirement Benefits (PRB) Other than Pensions (JUL 2005)
- 52.215-19 Notification of Ownership Changes (OCT 1997)
- 52.215-20 Requirements for Certified Cost or Pricing Data and Data Other than Certified Cost or Pricing Data (OCT 2010)
- 52.215-21 Requirements for Certified Cost or Pricing Data and Data Other than Certified Cost or Pricing Data Modifications (OCT 2010)
- 52.219-8 Utilization of Small Business Concerns (OCT 2014)
- 52.219-9 Small Business Subcontracting Plan (OCT 2014) Alternate II (OCT 2001)
- 52.219-16 Liquidated Damages—Subcontracting Plan (JAN 1999)
- 52.219-28 Post-Award Small Business Program Rerepresentation (JUL 2013)
- 52.222-1 Notice to the Government of Labor Disputes (FEB 1997)
- 52.222-3 Convict Labor (JUN 2003)
- 52.222-4 Contract Work Hours and Safety Standards Act—Overtime Compensation (MAY 2014)
- 52.222-6 Construction Wage Rate Requirements (MAY 2014)





- 52.222-7 Withholding of Funds (MAY 2014)
- 52.222-8 Payrolls and Basic Records (MAY 2014)

OBAYASHI

- 52.222-9 Apprentices and Trainees (JUL 2005)
- 52.222-10 Compliance with Copeland Act Requirements (FEB 1988)
- 52.222-11 Subcontracts (Labor Standards) (MAY 2014)
- 52.222-12 Contract Termination Debarment (MAY 2014)
- 52.222-13 Compliance with Construction Wage Rage Requirements and Related Regulations (MAY 2014)
- 52.222-14 Disputes Concerning Labor Standards (FEB 1988)
- 52.222-15 Certification of Eligibility (MAY 2014)
- 52.222-21 Prohibition of Segregated Facilities (APR 2015)
- 52.222-26 Equal Opportunity (APR 2015)
- 52.222-27 Affirmative Action Compliance Requirements for Construction (APR 2015)
- 52.222-30 Construction Wage Rate Requirements Price Adjustment (None or Separately Specified Method) (MAY 2014)
- 52.222-35 Equal Opportunity for Veterans (JUL 2014)
- 52.222-36 Equal Opportunity for Workers with Disabilities (JUL 2014)
- 52.222-37 Employment Reports on Veterans (JUL 2014)
- 52.222-40 Notification of Employee Rights Under the National Labor Relations Act (DEC 2010)
- 52.222-50 Combating Trafficking in Persons (MAR 2015)
- 52.222-54 Employment Eligibility Verification (AUG 2013)
- 52.222-55 Minimum Wages Under Executive Order 13658 (DEC 2014)
- 52.222-99 Establishing a Minimum Wage for Contractors (Deviation 2014- O0017) (JUN 2014)
- 52.223-2 Affirmative Procurement of Biobased Products Under Service and Construction Contracts (SEP 2013)
- 52.223-3 Hazardous Material Identification and Material Safety Data (JAN 1997)
- 52.223-5 Pollution Prevention and Right-To-Know Information (MAY 2011)
- 52.223-6 Drug-Free Workplace (MAY 2001)
- 52.223-9 Estimate of Percentage of Recovered Material Content for EPA Designated Products (MAY 2008)
 Alternate I (MAY 2008)
- 52.223-15 Energy Efficiency in Energy-Consuming Products (DEC 2007)
- 52.223-17 Affirmative Procurement of EPA-Designated Designed Items in Service and Construction Contracts (MAY 2008)
- 52.223-18 Encouraging Contractor Policies to Ban Text Messaging While Driving (AUG 2011)
- 52.224-1 Privacy Act Notification (APR 1984)
- 52.224-2 Privacy Act (APR 1984)
- 52.225-11 Buy American Construction Materials Under Trade Agreements (MAY 2014) Alternate I (MAY 2014)
- 52.225-13 Restrictions on Certain Foreign Purchases (JUN 2008)
- 52.227-1 Authorization and Consent (DEC 2007) Alternates I and II (APR 1984)
- 52.227-2 Notices and Assistance Regarding Patent & Copyright Infringement (DEC 2007)
- 52.227-4 Patent Indemnity Construction Contracts (DEC 2007) Alternate I (DEC 2007)
- 52.228-5 Insurance—Work on a Government Installation (JAN 1997)
- 52.229-3 Federal, State, and Local Taxes (FEB 2013)
- 52.232-5 Payments Under Fixed-Price Construction Contracts (MAY 2014)
- 52.232-17 Interest (MAY 2014)
- 52.232-23 Assignment of Claims (MAY 2014) Alternate I (APR 1984)
- 52.232-27 Prompt Payment for Construction Contracts (MAY 2014)
- 52.232-33 Payment by Electric Funds Transfer System for Award Management (JUL 2013)
- 52.232-39 Unenforceability of Unauthorized Obligations (JUN 2013)
- 52.232-40 Providing Accelerated Payments to Small Business Subcontractors (DEC 2013)
- 52.233-1 Disputes (MAY 2014) Alternate I (DEC 1991)
- 52.233-4 Applicable Law for Breach of Contract Claim (OCT 2004)
- 52.236-1 Performance of Work by the Contractor (APR 1984)
- 52.236-2 Differing Site Conditions (APR 1984)
- 52.236-3 Site Investigation and Conditions Affecting the Work (APR 1984)
- 52.236-4 Physical Data (APR 1984)





Granite - Obayashi, a Joint Venture

Reference No.: **733700** Subcontract No.: **08.0040** Page 25 of 31

52.236-5 Material and Workmanship (APR 1984)

OBAYASHI

52.236-7 Permits and Responsibilities (NOV 1991)

52.236-8 Other Contracts (APR 1984)

52.236-9 Protection of Existing Vegetation, Structures, Equipment, Utilities, and Improvements (APR 1984)

52.236-10 Operations and Storage Areas (APR 1984)

52.236-11 Use and Possession Prior to Completion (APR 1984)

52.236-12 Cleaning Up (APR 1984)

52.236-13 Accident Prevention (NOV 1991) Alternate I (NOV 1991)

52.236-14 Availability and Use of Utility Services (APR 1984)

52.236-15 Schedules for Construction Contracts (APR 1984)

52.236-16 Quantity Surveys (APR 1984)

52.236-17 Layout of Work (APR 1984)

52.236-21 Specifications and Drawings for Construction (FEB 1997) Alternates I and II (APR 1984)

52.242-13 Bankruptcy (JUL 1995)

52.242-14 Suspension of Work (APR 1984)

52.243-4 Changes (JUN 2007)

52.243-6 Change Order Accounting (APR 1984)

52.244-6 Subcontracts for Commercial Items (APR 2015)

52.245-1 Property Records (APR 2012), Alternate I (APR 2012)

52.245-9 Use and Charges (APR 2012)

52.246-12 Inspection of Construction (AUG 1996)

52.246-21 Warranty of Construction (MAR 1994) Alternate I (APR 1984)

52.248-3 Value Engineering – Construction (OCT 2010) Alternate I (APR 1984)

52.249-2 Termination for Convenience of the Government (Fixed-Price) (APR 2012) Alternate I (SEP 1996)

52.249-10 Default (Fixed-Price – Price Construction) (APR 1984)

52.251-1 Government Supply Sources (APR 2012)

52.253-1 Computer Generated Forms (JAN 1991)

DFARS CLAUSES INCORPORATED BY REFERENCE

- 252.203-7000 Requirements Relating to Compensation of Former DOD Officials (SEP 2011)
- 252.203-7001 Prohibition on Persons Convicted of Fraud or Other Defense-Contract-Related Felonies (DEC 2008)
- 252.203-7002 Requirement to Inform Employees of Whistleblower Rights (SEP 2013)
- 252.203-7003 Agency Office of the Inspector General (DEC 2012)
- 252.203-7004 Display of Fraud Hotline Poster(s) (JAN 2015)
- 252.203-7999 Prohibition on Contracting with Entities that Require Certain Internal Confidentiality Agreement (Deviation 2015-O0010) (FEB 2015)

252.204-7000 Disclosure of Information (AUG 2013)

252.204-7003 Control of Government Personnel Work Product (APR 1992)

252.204-7005 Oral Attestation of Security Responsibilities (NOV 2001)

252.204-7012 Safeguarding of Unclassified Controlled Technical Information (NOV 2013)

252.205-7000 Provisions of Information to Cooperative Agreement Holders (DEC 1991)

252.209-7004 Subcontracting with Firms that are Owned or Controlled by the Government of a Terrorist

Country (DEC 2014)

252.211-7003 Item Unique Identification and Valuation (DEC 2013)

252.211-7007 Reporting of Government-Furnished Property (AUG 2012)

252.215-7000 Pricing Adjustments (DEC 2012)

252.222-7000 Restrictions on Employment of Personnel (MAR 2000)

252.222-7006 Restrictions on the Use of Mandatory Arbitration Agreements (DEC 2010)

252.223-7001 Hazard Warning Labels (DEC 1991)

252.223-7004 Drug-Free Workplace (SEP 1988)

252.223-7006 Prohibition on Storage and Disposal of Toxic and Hazardous Materials (SEP 2014)





Granite – Obayashi, a Joint Venture

Reference No.: 733700 Subcontract No.: 08.0040 Page 26 of 31

252.223-7008	Prohibition of Hexavalent Chromium (JUN 2013)			
252.225-7005	Identification of Expenditures in the United States (JUN 2005)			
252.225-7012	Preference for Certain Domestic Commodities (FEB 2013)			
252.225-7041	Correspondence in English (JUN 1997)			
252.225-7048	Export-Controlled Items (JUN 2013)			
252.225-7993	Prohibition on Contracting with the Enemy (Deviation 2014-O0020) (SEP 2014)			
252.227-7022	Government Rights (Unlimited) (MAR 1979)			
252.227-7023	Drawings and Other Data to Become the Property of the Government (MAR 1979)			
252.227-7033	Rights in Shop Drawings (APR 1966)			
252.231-7000	Supplemental Cost Principles (DEC 1991)			
252.232-7003	Electronic Submission of Payment Requests and Receiving Reports (JUN 2012)			
252.232-7006	Wide Area Workflow Payment Instructions (MAY 2013)			
252.232-7010	Levies on Contract Payments (DEC 2006)			
252.236-7001				
252.236-7013 Requirement for Competition Opportunity for American Steel Producers, Fabricators, and				
	Manufacturers (JUN 2013)			
252.243-7001	£ , ,			
	Requests for Equitable Adjustment (DEC 2012)			
	Subcontracts for Commercial Items (JUN 2013)			
252.245-7001				
252.245-7002				
252.245-7003				
	Reporting, reutilization, and Disposal (MAR 2015)			
	Notification of Potential Safety Issues (JUN 2013)			
252.246-7004				
252.247-7023				
252.247-7024	Notification of Transportation of Supplies by Sea (MAR 2000)			

NFAS CLAUSES INCORPORATED BY REFERENCE

5252.201-9300	Contracting Officer Authority (JUN 1994)
5252.209-9300	Organizational Conflicts of Interest (JUN 1994)
5252.228-9300	Individual Surety/Sureties (JUN 1994)
	Accident Prevention (NOV 1998)
5252.236-9305	Availability of Utilities (JUN 1994)
5252.236-9310	Record Drawings (OCT 2004) Alternate I (JUN 1994) Alternate II (OCT 2004)
5252.242-9300	Government Representatives (OCT 1996)
5252.245-9302	Limited Assumption of Risk by Government (JUN 1994)

--End Document Table of Contents--





Reference No.: **733700** Subcontract No.: **08.0040** Page 27 of 31

DOCUMENT 00700 CONTRACT CLAUSES 06/15

1.1 FAR 52.252-2, CLAUSES INCORPORATED BY REFERENCE (FEB 1998)

This contract incorporates one or more clauses by reference, with the same force and effect as if they were given in full text. Upon request, the Contracting Officer will make their full text available. Also, the full text of a clause may be accessed electronically at this address:

DOCUMENT 00800 LABOR

REQUIREMENTS

1.1 NOTICE CONCERNING LABOR RATES

The Construction Wage Requirements statute (formerly known as the Davis-Bacon Act) will apply. The attached Guam General Wage Decision No. GU190001, Modification No. 0, Publication Date 01/04/2019 will remain in effect for the duration of the project.

Federal Acquisition Regulation (FAR) 22-404-6(c) requires incorporation of a wage determination received by the contracting office prior to award. If the contracting office receives a modification to the wage determination after receipt of proposals but prior award, at the discretion of the contracting officer, award of a contract resulting from this solicitation may be made without incorporation of the current wage determination prior to award and a modification will be issued after award to incorporate the wage determination effective as of the date of contract award.





Reference No.: **733700** Subcontract No.: **08.0040** Page 28 of 31

General Decision Number: GU190001 01/04/2019 GU1

Superseded General Decision Number: GU20180002

State: Guam

Construction Type: Heavy

Heavy Construction (Applies only to projects funded under the National Defense Authorization Act of 2010 - Guam Realignment Fund - Defense Policy Review Initiative)

County: Guam Statewide.

Note: Under Executive Order (EO) 13658, an hourly minimum wage of \$10.60 for calendar year 2019 applies to all contracts subject to the Davis-Bacon Act for which the contract is awarded (and any solicitation was issued) on or after January 1, 2015. If this contract is covered by the EO, the contractor must pay all workers in any classification listed on this wage determination at least \$10.60 per hour (or the applicable wage rate listed on this wage determination, if it is higher) for all hours spent performing on the contract in calendar year 2019. If this contract is covered by the EO and a classification considered necessary for performance of work on the contract does not appear on this wage determination, the contractor must pay workers in that classification at least the wage rate determined through the conformance process set forth in 29 CFR 5.5(a)(1)(ii) (or the EO minimum wage rate, if it is higher than the conformed wage rate). The EO minimum wage rate will be adjusted annually. Please note that this EO applies to the above-mentioned types of contracts entered into by the federal government that are subject to the Davis-Bacon Act itself, but it does not apply to contracts subject only to the Davis-Bacon Related Acts, including those set forth at 29 CFR 5.1(a)(2)-(60). Additional information on contractor requirements and worker protections under the EO is available at www.dol.gov/whd/govcontracts.

Modification Number Publication Date 0 01/04/2019

SUGU2016-003 02/08/2017

F	Rates	Fringes
CARPENTER (Form Work Only)\$	17.04	0.00
CEMENT MASON/CONCRETE FINISHER\$	12.87	0.00
ELECTRICIAN\$	15.20	0.00
IRONWORKER, REINFORCING\$	12.56	0.00
LABORER: Common or General\$	10.36	0.00







Reference No.: 733700 Subcontract No.: 08.0040 Page 29 of 31

OPERATOR:

 Backhoe/Excavator/Trackhoe.....\$ 14.44
 0.00

 OPERATOR: Mechanic......\$ 16.66
 0.00

 TRUCK DRIVER: Dump Truck......\$ 13.82
 0.00

WELDERS - Receive rate prescribed for craft performing operation to which welding is incidental.

Note: Executive Order (EO) 13706, Establishing Paid Sick Leave for Federal Contractors applies to all contracts subject to the Davis-Bacon Act for which the contract is awarded (and any solicitation was issued) on or after January 1, 2017. If this contract is covered by the EO, the contractor must provide employees with 1 hour of paid sick leave for every 30 hours they work, up to 56 hours of paid sick leave each year. Employees must be permitted to use paid sick leave for their own illness, injury or other health-related needs, including preventive care; to assist a family member (or person who is like family to the employee) who is ill, injured, or has other health-related needs, including preventive care; or for reasons resulting from, or to assist a family member (or person who is like family to the employee) who is a victim of, domestic violence, sexual assault, or stalking. Additional information on contractor requirements and worker protections under the EO is available at www.dol.gov/whd/govcontracts.

Unlisted classifications needed for work not included within the scope of the classifications listed may be added after award only as provided in the labor standards contract clauses (29CFR 5.5 (a) (1) (ii)).

The body of each wage determination lists the classification and wage rates that have been found to be prevailing for the cited type(s) of construction in the area covered by the wage determination. The classifications are listed in alphabetical order of "identifiers" that indicate whether the particular rate is a union rate (current union negotiated rate for local), a survey rate (weighted average rate) or a union average rate (weighted union average rate).

Union Rate Identifiers

A four letter classification abbreviation identifier enclosed in dotted lines beginning with characters other than "SU" or "UAVG" denotes that the union classification and rate were prevailing for that classification in the survey. Example: PLUM0198-005 07/01/2014. PLUM is an abbreviation identifier of the union which prevailed in the survey for this classification, which in this example would be Plumbers. 0198 indicates the local union number or district council number





Reference No.: 733700 Subcontract No.: 08.0040 Page 30 of 31

where applicable, i.e., Plumbers Local 0198. The next number, 005 in the example, is an internal number used in processing the wage determination. 07/01/2014 is the effective date of the most current negotiated rate, which in this example is July 1, 2014.

Union prevailing wage rates are updated to reflect all rate changes in the collective bargaining agreement (CBA) governing this classification and rate.

Survey Rate Identifiers

Classifications listed under the "SU" identifier indicate that no one rate prevailed for this classification in the survey and the published rate is derived by computing a weighted average rate based on all the rates reported in the survey for that classification. As this weighted average rate includes all rates reported in the survey, it may include both union and non-union rates. Example: SULA2012-007 5/13/2014. SU indicates the rates are survey rates based on a weighted average calculation of rates and are not majority rates. LA indicates the State of Louisiana. 2012 is the year of survey on which these classifications and rates are based. The next number, 007 in the example, is an internal number used in producing the wage determination. 5/13/2014 indicates the survey completion date for the classifications and rates under that identifier.

Survey wage rates are not updated and remain in effect until a new survey is conducted.

Union Average Rate Identifiers

Classification(s) listed under the UAVG identifier indicate that no single majority rate prevailed for those classifications; however, 100% of the data reported for the classifications was union data. EXAMPLE: UAVG-OH-0010 08/29/2014. UAVG indicates that the rate is a weighted union average rate. OH indicates the state. The next number, 0010 in the example, is an internal number used in producing the wage determination. 08/29/2014 indicates the survey completion date for the classifications and rates under that identifier.

A UAVG rate will be updated once a year, usually in January of each year, to reflect a weighted average of the current negotiated/CBA rate of the union locals from which the rate is based.

WAGE DETERMINATION APPEALS PROCESS

- 1.) Has there been an initial decision in the matter? This can be:
- * an existing published wage determination
- * a survey underlying a wage determination
- * a Wage and Hour Division letter setting forth a position on a wage determination matter





Reference No.: **733700** Subcontract No.: **08.0040** Page 31 of 31

* a conformance (additional classification and rate) ruling

On survey related matters, initial contact, including requests for summaries of surveys, should be with the Wage and Hour Regional Office for the area in which the survey was conducted because those Regional Offices have responsibility for the Davis-Bacon survey program. If the response from this initial contact is not satisfactory, then the process described in 2.) and 3.) should be followed.

With regard to any other matter not yet ripe for the formal process described here, initial contact should be with the Branch of Construction Wage Determinations. Write to:

> Branch of Construction Wage Determinations Wage and Hour Division U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

2.) If the answer to the question in 1.) is yes, then an interested party (those affected by the action) can request review and reconsideration from the Wage and Hour Administrator (See 29 CFR Part 1.8 and 29 CFR Part 7). Write to:

> Wage and Hour Administrator U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

The request should be accompanied by a full statement of the interested party's position and by any information (wage payment data, project description, area practice material, etc.) that the requestor considers relevant to the issue.

3.) If the decision of the Administrator is not favorable, an interested party may appeal directly to the Administrative Review Board (formerly the Wage Appeals Board). Write to:

Administrative Review Board U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

4.) All decisions by the Administrative Review Board are final.

END OF GENERAL DECISION





Exhibit "C"

EXHIBIT C



Name of Contract: Address on Contract:

RELYANT Global, LLC.				
335 High Street				
Maryville, TN 37804				

Invoice Number Paid when Paid Plus 7 Date of invoic G/15/2023 RG Project Ref: 20147 7ARY5

CAGE Code:

Partial Invoice

Attention to:

Client:

Prime Contract Number: PO/SubContract Project Number/Name: Invoice - Period of Performance Granite-Obayashi Joint Venture 425 Chalan San Antonio PMB #1023 Tamuning, GU 96913 Aaron Snyder, Project Director

N62742-17-C-1324 08.0040

J-001B Finegayan Utilities and Site Improvements Phase I 01/01/2023 -05/05/2023

	Description	100	Unit	Qty/Unit - Completed to Date			Performance				
Item		Qty		-		400				Employee The Involve	
	Project Planing and Admin		LS	Prov Inv.	This Invoice	To Date	% Comp 100.00%	5	Contract Price 16,723.80	Amount Due - I his invoice	Total Invoiced to Date 5 16,723.
2	MEC Work Plan Development	1	LS	1.00	0	1.00	100.00%	5	8,205.50		\$ 8,205.
1	Mobilization/Demobilization	1	LS	1.00	0	1.00	100.00%	ć	95,754.90		\$ 95,754.
4	Surface Clearance - Area F	28	AC	25.00	0	28.00	100.00%	1	64,621.20		\$ 64,621.
5	Anomaly Investigation & EZ Controls - Area F	28	LS	28.00	0	28.00	100.00%	5	929.362.84		5 929.362
6	After Action Report	2	EA	1.00	1	200	100.00%	5	10,344.00	\$ 5,172.00	\$ 10,344.
7	AA Support / Spotter	5540	HR	5,540,00	0	5,540,00	100.00%	5	592,281.40	\$ 0,172.00	5 592.281.
8	Bond	1	LS	1.00	0	1.00	100.00%	5	18,890.23		\$ 18,890.
IANGE ORI				1.00		130	100.0078	1	10,070.23		10,070.
CO I.I	Spread & Scans	106689	SM	104,689,00	0	106,689.00	100.00%	5	84,284.31		\$ 84,284.
CO 1.1	E1-16B for 150' radius of 5" Projectile	6567	SM	6,567.00	0	6,567.00	100.00%	5	11,032.56		\$ 11,032
CO 2.2a	Surface Clearance - Area A	15	AC	15.00	0	15.00	100.00%	5	34,618.50		\$ 34,618.
CO 2.2b	Anomaly Investigation & EZ Controls - Area A	15	AC	15.00	0	15.00		5	497.872.95		\$ 497,872
CO 23a	Surface Clearance - Area B	35	AC	35.00	0	35.00	100.00%	5	80.776.50		\$ 80,776
CO 23b	Anomaly Investigation & EZ Controls - Area B	35	AC	35.00	0	35.00	100.00%	5	1,161,703.55		
CO 24a	Surface Clearance - Area D	35	AC	35.00	0	35.00	100.00%	5	80.776.50		\$ 1,161,703. \$ 80,776.
					_			-			
CO 24b	Anomaly Investigation & EZ Controls - Area D	35	AC	34.61	0	34.61	98.89%	\$	1,161,703.55	e 00.000.00	\$ 1,148,758.
	After Action Report	4	EA	0.00	4	4.00	100.00%	\$	20,688.00	-	\$ 20,688.
CO 2.6	AA Support / Spotter	14910	HR	6,826.50	61	6,887.50	46.19%	\$	1,594,028.10	\$ 6,521.51	\$ 736,342
CO 27	Bond		LS	1.00	0	1.00	100.00%	5	50,953.85		\$ 50,953
CO 3	Add'l SUXOS & QCSO Nights for Trash Pits	10	DAY	10.00	0	10.00	100.00%	\$	26,360.40		\$ 26,360.
CO 4.1	Spread & Scans	14259	SM	14,259.00	0	14,259.00	100.00%	\$	11,264.61		\$ 11,264
CO 42	E1-168 for South Basin 5	1252	MZ	1,252.00	0	1,252.00	100.00%	\$	2,103.36		\$ 2,103.
CO 5	E1-16B for Topsoil Stockpiles	25000	CY	25,000.00	D	25,000.00	100.00%	\$	70,250.00		\$ 70,250.
CO 6.1	Add'l Project Planing and Admin (PCO-17)	1	LS	1.00	0	1.00	100.00%	5	836.19		\$ 836.
CO 6.2	Work Plan Addendum (If required) (PCO-17)	1	LS	0.00	0	0.00	0%	\$	410.27		
CO 63	Surface Clearance (PCO-17)	3.55	AC	3.55	0	3.55	100.00%	\$	8,193.05		\$ 8,193.
CO 6.4	Anomaly Investigation & EZ Controls (PCO-17)	1	LS	1.00	0	1.00	100.00%	\$	153,128.19		\$ 153,128.
CO 65	Bond	ı	LS	1.00	0	1.00	100.00%	\$	1,820.76		\$ 1,820.
CO 7.1	E1-16B Scans at Area 4E	22621	SM	22,621.21	D	22,621.21	100.00%	\$	38,003.63		\$ 38,003.
CO 72	Spread and Scan of Original Contract at Area 4E	30190	MZ	30,189.59	0	30,189_59	100.00%	5	23,849.78		\$ 23,849.
CO 7.3	E1-16B Scans for Sinkhole Exploration at Area B.	201	MZ	201.00	0	201.00	100.00%	5	337.68		\$ 337.
CO 7.4	AA Support/Spotter for Sinkhole Exploration at Area B	20	HR	20.00	0	20.00	100.00%	\$	2,138.20		\$ 2,138.
CO 8	E1-16B for Topsoil Stockpiles	60000	CY	60,000.00	D	60,000.00	100.00%	\$	168,600.00		\$ 168,600.
CO 9.1	Additional SUXOS & QC/SO at Day shift at 4A/4H Stockpile	В	DY	8.00	0	8.00	100.00%	\$	21,088.32		\$ 21,088.
CO 9.2	Surface Clearance at Smart Growth	26	AC	26.00	0	26.00	100.00%	\$	60,005.40		\$ 60,005.
CO 10.1	Mechanical Trommel Screening Operations First Trommel	100	HR	100.00	0	100.00	100.00%	\$	61,271.00		\$ 61,271.
CO 10.2	Each Additional Trommel's Operations	100	HR	100.00	0	100.00	100.00%	\$	22,856.00		\$ 22,856.
CO II.I	Amend. of CO 9.2 Surface Clearance at Smart Growth	-26	AC	-26.00	0	-26.00	100.00%	5	(60,005.40)		\$ (60,005.
CO 11.2	Amend. of CO 9.2 Surface Clearance at Smart Growth	11.73	AC	11.73	0	11.73	100.00%	5	27,071.67		\$ 27,071.
CO 12.1	Anomaly Investigation, Surface Clearance & EZ Control/ Signage	-3.5	AC	-3.50	0	-3.50	100.00%	5	(124,248.01)		5 (124,248.
CO 122	Topsoil Scanning	157276	SM	157,276.00	0	157,276.00	100.00%	5	124,248.04		\$ 124,248.
CO 13.1	First Trommel's Operation minimum 40hrs per week	48	HR	48.00	0	48.00	100.00%	5	29,410.08		\$ 29,410.
CO 13.2	Each Additional Trommel's Operations	-100	HR	-100.00	D	-100.00	100.00%	5	(22,856.00)		\$ (22,856.
CO 14.1	E1-16B for Topsoil Stockpiles (Unsigned/Partial)	50000	CY	50,000.00	0	50,000.00	100.00%	5	140,500.00		\$ 140,500
CO 14.2	Additional Surface Clearance at Smart Growth (Unsigned/Partial)	10	AC	5.90	0,000	5.8%	58.96%	5	23.079.00		5 13,607
CO 15	Mechanical Trommel Screen WP	1	LS	1.00	0	1.00	100.00%	5	10.477.88		5 10,477
CO 16	E1-16B for Topsoil Stockpiles (Unsigned/Partial)	60000	CY	60,000	0	60,000	100.00%	5	168,600.00		\$ 168,600
CO 19	E1-16B for Topsoil Stockpiles overage	2851.13	CY	2,851.13	0		100.00%	5	8,011.68		\$ 8,011
_			_	_		2,851.13		-			
CO 21	Topsoil Spread and Scan TOI 20mm	12,000	SM	12,000.00	0	12,000.00	100.00%	5	27,120.00		\$ 27,120
CO 22	Topsoil Spread and Scan TCH 20mm	15,000	MZ	14,992.55	0	14,992.55	99.95%	5	33,900.00		\$ 33,883
CO 23	Topsoil Spread and Scan TOI 20mm	30,000	MZ	0.00	29,996.46	29,998.48	99.99%	5	67,800.00	\$ 67,796.56	\$ 67,796
		-		-	-		SubTotal	\$	7,640,248.02	\$ 100,178.07	\$ 6,759,713.
						Less Amount Prev	iousch Rillad				\$ 6,659,535.
						CLAST SHIP WILL I'V	DUCASING DESIGNATION				4 0,000,000

100,178.07 Net Amount Due this Invoice % Contract Completed

Company Point of Contact

Jennifer Goodpastu Projects Control

ACH Routing Number: SWIFT Code: Bank Name:



Exhibit "D"



24 February 2023 **Submitted via email**

Kenneth.McClain@gcinc.com

Granite Obayashi JV Attn: Kenneth McClain 425 Chalan San Antonio PMB #1023 Tamuning, GU 96913

REFERENCE: CONTRACT NO. N62742-17-C-1324, SUBCONTRACT NO. 08.0040,

RG20147 GRANITE FINEGAYAN PHASE 1, CAMP BLAS, GUAM

SUBJECT: Request for Equitable Adjustment

16B Soil Piles

Mr. McClain,

Please accept this correspondence as RELYANT Global, LLC's ("RELYANT") request for equitable adjustment of the referenced Subcontract for the cost associated with the high saturation of anomalies in the 16B Soil Piles in areas 4H and 4A.

Pursuant to Section 14.4 of Subcontract No. 08.0040 ("Subcontract"), in the event that RELYANT encounters any condition or event during the performance of Work that requires an adjustment in the Subcontract price and/or the Work Progress Schedule ("Site Condition"), RELYANT is entitled to an adjustment in compensation to the extent a commensurate adjustment in compensation or time is allowed to GOJV by NAVFAC Marianas ("Government") for costs incurred by RELYANT that were caused by the Site Condition. Further, the Subcontract incorporates 48 CFR § 52.236-2 – Differing Site Conditions, by reference, under which an equitable adjustment of the Subcontract shall be made where a subsurface or latent physical condition at the Site differs materially from those indicated in the Subcontract or where unknown physical conditions at the Site, of an unusual nature, which differ materially from those ordinarily encountered and generally recognized as inhering work of the character provided for in the Contract. Similarly, pursuant to Section 14.0 of the Subcontract, all changes ordered by GOJV, but which originate with or are requested by the NAVFAC Marianas, will be compensated for by GOJV to the extent a commensurate adjustment in compensation or time is allowed GOJV by NAVFAC Marianas.

I. RELYANT based its pricing on the subsurface conditions indicated in the Subcontract Documents, which provided an anomaly detection rate of 600 to 1,800 per acre.

On 14 August 2020, RELYANT submitted a final revised offer in response to Granite Obayashi JV's ("GOJV") request for proposal for MEC work to be carried out for Contract No. N62742-17-C-1324 (the "Contract"). The Contract Request for Proposal and the Subcontract (collectively the "Subcontract Documents") required compliance with Joint Region Marianas Explosive Safety Submission included CNO Exemption E1-16B Chief of Naval Operations Explosives Safety Exemption E1-E16A dated 05 January 2017, Serial Letter N41/17U129836 ("CNO Exemption"), which states that the "current anomaly detection rate in Guam varies from 600 to 1,800 per acre." Additionally, the 16B soil piles were subject to NAVFAC MEC QA during the CNO Exemption clearance process in which they were deemed "Yellow" soils via QA



acceptance. For these reasons, RELYANT anticipated that the 16B soil piles contained minimal ferrous anomalies.

Accordingly, RELYANT relied on the Subcontract Documents to calculate its proposed pricing based on an estimated anomaly count not to exceed 1,800 anomalies per acre. Notably, the bid schedule GOJV provided to RELYANT did not include a separate CLIN for MEC anomalies per acre. Therefore, RELYANT could not submit costs for additional MEC anomalies per acre.

II. The 16B soil piles differed materially from those indicated in the Subcontract Documents.

The CNO Exemption provided in the Subcontract Documents, which RELYANT relied on as a basis for its proposed pricing, is flawed or was not previously followed. To date, RELYANT encountered significant anomaly density (e.g. cultural debris, construction debris, MEC items, munitions debris, etc.) throughout the 16B soil piles, which cover an area of approximately 9.14 acres. The anomaly count in these areas exceeded the anticipated average per acre anomaly count. On average, RELYANT encountered 7,779.6 anomalies per acre within the areas defined, well over the maximum anomaly rate contained in the JRM ESS specifications. RELYANT also removed a total of 25,223 pounds of cultural debris, 510 pounds of MDAS, and seven (7) MEC items in this area.

Simply put, were the information provided in the Subcontract Documents accurate, significant MEC, munitions debris, construction debris, and cultural debris would not be present in the 16B soil piles. Rather, the 16B soil piles should have limited or minimal anomalies.

III. RELYANT requests the Subcontract be equitably adjusted to account for the cost incurred as a result of the high saturation of anomalies in 16B Soil Piles in areas 4H and 4A.

In *Int'l Tech. Corp. v. Winter*, 523 F.3d 1341 (Fed. Cir. 2008), the United States Court of Appeals for the Federal Circuit discussed the legal standard applicable to differing site conditions. In *Int'l Tech. Corp.*, the court addressed a contractor's DSC claim for additional costs arising out of the contractor's treatment of contaminated soil encountered during the course of the project. The Court articulated the test applicable to the claim and identified four essential elements for establishing a *prima facie* claim. First, the contractor must prove that a reasonable contractor reading the contract documents would interpret them as making a representation as to the existing site conditions. Second, the contractor must prove that the actual site conditions were not reasonably foreseeable to the contractor based on information available to the contractor outside of the contract documents such that the contractor's reliance on the indications in the contract was reasonable. Third, the contractor must prove that it in fact relied on the contract indications. Fourth, the contractor must prove that the conditions differed materially from those represented and that the contractor suffered damages as a result.

Here, RELYANT relied on the CNO Exemption contained in the Contract Documents, which indicated an anomaly rate of 600 to 1,800 anomalies per acre. Therefore, nothing contained in the Contract Documents, nor any information made available to RELYANT outside of the Contract Documents, gave rise to any reasonable assumption that anomaly rate provided in the Contract Documents were flawed. In fact, the soils were deemed "Yellow." Acting in reasonable reliance on the Contract Documents, RELYANT provided its bid based on an anomaly rate of 600 to 1,800 anomalies per acre and planned its operations accordingly. Specifically, RELYANT estimated that production rates for two teams consisting of 1x SUXOS, 1x QC, 1x Safety, 2x Team Leaders, 4x Tech II, 4x Tech I, and 2x HEO Operators would be two (2) acres every



four (4) days working ten (10) hour days. Upon beginning work, RELYANT learned that the anomaly rate in the 16B soil piles is, in fact, significantly higher than that provided in the Contract Documents. Please reference Figure One below, which demonstrates anomaly expected versus anomalies encountered.

	Expected	Actual			
Anomaly	1,800 per acre	7,779.6 per acre			

Figure One

The foregoing caused RELYANT to incur an additional \$1,658,036.77 in cost as quantified in Attachment A, SF 4330 and associated cost details. Therefore, RELYANT requests the Subcontract be increased by \$1,658,036.77. RELYANT possesses documentation to support all costs in this request. If required, please advise a transmission link to upload said documentation as it is too large to transmit via email.

In accordance with 48 CFR § 252.243-7002, incorporated into the Subcontract by reference, I certify that this request is made in good faith, and that the supporting data are accurate and complete to the best of my knowledge and belief.

Respectfully,

Shelby C. Burchell
Shelby C. Burchell

Director of Contracts & Compliance

RELYANT Global, LLC

Attachments:

Attachment A: SF 4330 and associated cost details

Exhibit "E"



24 February 2023 **Submitted via email**

Kenneth.McClain@gcinc.com

Granite Obayashi JV Attn: Kenneth McClain 425 Chalan San Antonio PMB #1023 Tamuning, GU 96913

REFERENCE: CONTRACT NO. N62742-17-C-1324, SUBCONTRACT NO. 08.0040,

RG20147 GRANITE FINEGAYAN PHASE 1, CAMP BLAS, GUAM

SUBJECT: Request for Equitable Adjustment

Area F and F Extension Trash Pits

Mr. McClain,

Please accept this correspondence as RELYANT Global, LLC's ("RELYANT") request for equitable adjustment of the referenced Subcontract for the cost associated with the high saturation of anomalies in the Area F and F Extension Trash Pits.

Pursuant to Section 14.4 of Subcontract No. 08.0040 ("Subcontract"), in the event that RELYANT encounters any condition or event during the performance of Work that requires an adjustment in the Subcontract price and/or the Work Progress Schedule ("Site Condition"), RELYANT is entitled to an adjustment in compensation to the extent a commensurate adjustment in compensation or time is allowed to GOJV by NAVFAC Marianas ("Government") for costs incurred by RELYANT that were caused by the Site Condition. Further, the Subcontract incorporates 48 CFR § 52.236-2 – Differing Site Conditions, by reference, under which an equitable adjustment of the Subcontract shall be made where a subsurface or latent physical condition at the Site differs materially from those indicated in the Subcontract or where unknown physical conditions at the Site, of an unusual nature, which differ materially from those ordinarily encountered and generally recognized as inhering work of the character provided for in the Contract. Similarly, pursuant to Section 14.0 of the Subcontract, all changes ordered by GOJV, but which originate with or are requested by the NAVFAC Marianas, will be compensated for by GOJV to the extent a commensurate adjustment in compensation or time is allowed GOJV by NAVFAC Marianas.

I. RELYANT based its pricing on the subsurface conditions indicated in the Subcontract Documents, which provided an anomaly detection rate of 600 to 1,800 per acre.

On 14 August 2020, RELYANT submitted a final revised offer in response to Granite Obayashi JV's ("GOJV") request for proposal for MEC work to be carried out for Contract No. N62742-17-C-1324 (the "Contract"). The Contract Request for Proposal and the Subcontract (collectively the "Subcontract Documents") required compliance with Joint Region Marianas Explosive Safety Submission included CNO Exemption E1-16B Chief of Naval Operations Explosives Safety Exemption E1-E16A dated 05 January 2017, Serial Letter N41/17U129836 ("CNO Exemption"), which states that the "current anomaly detection rate in Guam varies from 600 to 1,800 per acre." For this reason, RELYANT did not anticipate that Area F



and F Extension contained ferrous anomalies.

Accordingly, RELYANT relied on the Subcontract Documents to calculate its proposed pricing based on an estimated anomaly count not to exceed 1,800 anomalies per acre. Notably, the bid schedule GOJV provided to RELYANT did not include a separate CLIN for MEC anomalies per acre. Therefore, RELYANT could not submit costs for additional MEC anomalies per acre.

II. The Area F and F Extension Trash Pits differed materially from those indicated in the Subcontract Documents.

The CNO Exemption provided in the Subcontract Documents, which RELYANT relied on as a basis for its proposed pricing, is flawed or was not previously followed. To date, RELYANT encountered significant anomaly density (e.g. cultural debris, construction debris, MEC items, munitions debris, etc.) throughout the Area F and F Extension Trash Pits. Please reference Figure One, EZ Map, below.



Figure One, EZ Map

The affected grids in Area F and F Extension, T5, T8, U5, U6, V5, V6, U10, and T10, total 4.73 combined



acres. The anomaly count in these areas exceeded the anticipated average per acre anomaly count. On average, RELYANT encountered 14,791.33 anomalies per acre within the areas defined, well over the maximum anomaly rate contained in the JRM ESS specifications. RELYANT also removed a total of 271,567 pounds of cultural debris, 319 pounds of MDAS, and three (3) MEC items in this area.

Simply put, were the information provided in the Subcontract Documents accurate, significant MEC, munitions debris, construction debris, and cultural debris would not be present in the Area F and F Extension Trash Pits. Rather, the Trash Pits should have limited or minimal anomalies.

III. RELYANT requests the Subcontract be equitably adjusted to account for the cost incurred as a result of the high saturation of anomalies in Areas F and F Extension Trash Pits.

In Int'l Tech. Corp. v. Winter, 523 F.3d 1341 (Fed. Cir. 2008), the United States Court of Appeals for the Federal Circuit discussed the legal standard applicable to differing site conditions. In Int'l Tech. Corp., the court addressed a contractor's DSC claim for additional costs arising out of the contractor's treatment of contaminated soil encountered during the course of the project. The Court articulated the test applicable to the claim and identified four essential elements for establishing a prima facie claim. First, the contractor must prove that a reasonable contractor reading the contract documents would interpret them as making a representation as to the existing site conditions. Second, the contractor must prove that the actual site conditions were not reasonably foreseeable to the contractor based on information available to the contractor outside of the contract documents such that the contractor's reliance on the indications in the contract was reasonable. Third, the contractor must prove that it in fact relied on the contract indications. Fourth, the contractor must prove that the conditions differed materially from those represented and that the contractor suffered damages as a result.

Here, RELYANT relied on the CNO Exemption contained in the Contract Documents, which indicated an anomaly rate of 600 to 1,800 anomalies per acre. Therefore, nothing contained in the Contract Documents, nor any information made available to RELYANT outside of the contract documents, gave rise to any reasonable assumption that anomaly rate provided in the Contract Documents were flawed. Acting in reasonable reliance on the Contract Documents, RELYANT provided its bid based on an anomaly rate of 600 to 1,800 anomalies per acre and planned its operations accordingly. Specifically, RELYANT estimated 1,520-man hours to perform excavations in grids T5, T8, U5, U6, V5, V6, U10, and T10, utilizing three lifts of 18-inches to a construction depth of five feet. Please reference RELYANT's Addendum #1 to the ICC Work Plan under Prime Contract N62742-17-C-1324.

Upon beginning work, RELYANT learned that the anomaly rate in the Area F and F Extension Trash Pits is, in fact, significantly higher than that provided in the Contract Documents. For example, in grids U5 and U6, RELYANT performed eight lifts of 18-inches rather than the originally planned three lifts of 18-inches. Please reference Figure Two, Expected vs. Actuals, below, which demonstrates anomaly and man hours expected versus anomalies encountered and man hours expended.



	Expected	Actual
Anomaly	1,800 per acre	14,791.33 per acre

Figure Two, Expected vs. Actuals

Demonstrative of RELYANT's effort to mitigate incurred cost, although RELYANT encountered roughly 8.2 times the expected number of anomalies per acre, it only expended roughly 6.2 times the number of hours originally planned. Nonetheless, the foregoing caused RELYANT to incur an additional \$1,101,355.53 in cost as quantified in Attachment A, SF 4330 and associated cost details. Therefore, RELYANT requests the Subcontract be increased by \$1,101,355.53. RELYANT possesses documentation to support all costs in this request. If required, please advise a transmission link to upload said documentation as it is too large to transmit via email.

In accordance with 48 CFR § 252.243-7002, incorporated into the Subcontract by reference, I certify that this request is made in good faith, and that the supporting data are accurate and complete to the best of my knowledge and belief.

Respectfully,

Shelby C. Burchell
Shelby C. Burchell

Director of Contracts & Compliance

RELYANT Global, LLC

Attachments:

Attachment A: SF 4330 and associated cost details